Housing Consumption and Bubble Size

Stefanie J. Huber
* Christina Rott \ddagger Giovanni Giusti \dagger

DRAFT: UNDER REVISION

Version: June 10, 2022

Abstract

This paper identifies a novel determinant of the magnitude of house price bubbles: the demand for housing consumption. We provide a theoretical framework and investigate the role of this factor with an innovative experimental design that allows modeling of the consumption side and the investment side of housing. The overlapping generation structure, as well as the endogenous nature of the dividend (rental income) and, thus, the endogenous fundamental value of the housing asset make our experimental design a credible framework for studying house price bubbles. The results show that the lower the demand for housing consumption, the smaller the fundamental value of housing and the larger the size of the house price bubble.

JEL Classifications: E00, C91, G10, G12

Keywords: Housing Bubbles, Bubble Size, Housing Consumption, Experimental Asset Markets, OLG models.

We would like to thank John Duffy, Jordi Galí, Luba Petersen, Rosemarie Nagel, Jona Linde, Martin Strobel, Shyam Sunder, and Matthias Weber for valuable comments and suggestions. We benefited from comments and suggestions received during presentations at the Barcelona GSE Summer Forum's Workshop on Theoretical and Experimental Macroeconomics, the 3^{rd} International Meeting on Experimental and Behavioral Social Sciences (IMEBESS), the 11^{th} Maastricht Behavioral and Experimental Economics Symposium (M-BEES), the 24^{th} International Conference Computing in Economics and Finance (CEF), the 1^{st} European Midwest Micro/Macro Conference (EM3C), and the 2019 Asian-Pacific ESA Meeting. We are especially grateful to Barcelona GSE and the Elinor Ostrom Research Grant by the Elinor Ostrom Fund from the SBE Maastricht University for the funding of this project. Any errors remain our own.

 $^{^{\}star}$ CeNDEF, Amsterdam School of Economics, University of Amsterdam and Tinbergen Institute, Amsterdam, the Netherlands. E-mail: s.j.huber@uva.nl.

[‡] Department of Management and Organization, School of Business and Economics, Vrije Universiteit Amsterdam and Tinbergen Institute, the Netherlands. E-mail: c.e.rott@vu.nl.

[†] Escola Superior de Ciències Socials i de l'Empresa (Tecnocampus, Universitat Pompeu Fabra), Spain. E-mail: ggiusti@tecnocampus.cat.

1 Introduction

House price fluctuations have a substantial impact on economic performance. Empirical studies show that recessions associated with house price busts are not only twice as long but also twice as deep compared to normal recessions or recessions associated with equity price busts (Claessens et al. (2012); Claessens et al. (2009); IMF (2003)). Understanding which factors determine the magnitude of housing bubbles is therefore essential. In this paper, we investigate whether the consumption side of the housing asset is a driving factor for the house price bubble size.¹

Housing is a particular asset given its dual nature. The housing asset has a consumption side (it provides housing services) and an investment side. Conceptually, it is crucial to separate the consumption from the investment side as housing consumption and housing investment combine to determine housing market outcomes. Both housing investment and housing consumption are economically important from a household perspective. Typically, investing in a housing asset constitutes the most significant investment of a household within its lifetime. Households consume housing services independently of whether they own or rent the property they are living in. The larger the property and/or the better its quality, the more housing services are consumed and the higher the utility from housing consumption. Housing consumption constitutes the single-highest household expenditure item in all OECD and EU countries (Eurostat, 2020). However, the relative importance of housing consumption varies however greatly across countries.

We shed new light on the role of housing consumption as a potential determinant of the magnitude of house price bubbles by testing Huber (2018)'s model prediction empirically. Comparative statics show that a weak (strong) preferences for housing services relative to other consumption goods lead to a low (high) demand for housing consumption, resulting in a low (high) share of consumption expenditure being spent on housing consumption as a model equilibrium outcome. Consequently, the dividend of the housing asset (i.e., price for housing services) and the fundamental value (FV) are expected to be smaller the weaker the preference for housing services. This in turn has crucial implications for the maximum bubble size: In equilibrium, every asset needs to be affordable. It follows from the resource constraint of the economy that an economy with a weak aggregate preference for housing services will have more room for larger bubbles.

The theoretical prediction that the demand for housing consumption (relative to other

¹Throughout the paper, we use "housing consumption" and "housing services" interchangeably. If not noted otherwise, we refer to "bubbles" as the differences between the trading price and the fundamental value of the asset (i.e., overpricing).

consumption goods) is a driving factor for the magnitude of housing bubbles needs to be validated empirically. In a cross-country comparison and over the period of 1970–2019, Huber (2018) shows for a sample of 18 OECD countries that countries characterized by lower housing consumption experience bubbles of significantly larger magnitude.² While this observational data has the advantage of providing realistic, real–world information, this data also has its limitations due to measurement problems and allows only for a correlational interpretation.³ A laboratory experiment can overcome both the measurement issues and the data limitations: In the lab, housing services, the fundamental value and traded house prices are known. Hence, one key advantage is that the bubble size can be measured without error. Further, the controlled environment allows us to establish a causal relationship.

To test the hypothesis that the demand for housing consumption (relative to other consumption goods) is a driving factor for the magnitude of housing bubbles, two treatments are implemented—one with a weak preference for housing services and one with a strong preference for housing services relative to other consumption goods. The weak and the strong preferences for housing services are randomly and exogenously induced across sessions by varying the corresponding parameter in the household's utility function. The key parameter, the preference for housing services, is calibrated to match the observational data (CPI weight on housing services, incl. imputed rents).

The design of the laboratory experiment is closely related to the overlapping generation (OLG) model of rational housing bubbles of Huber (2018), which extends the framework of Galí (2014). Following the main model features, the investment and consumption side of the housing asset is explicitly (and separately) modeled. Households live for two periods in the

²Housing consumption is measured by the CPI weight on housing services, including imputed rents. This aggregate indicator presents information on the final consumption expenditure of households on housing (incl. side costs such as water, electricity, gas, and other fuels), as a percentage of their overall final consumption expenditure. This indicator highlights the relative importance of housing-related expenditures within consumer spending and facilitates comparison with other household budget items across countries. In 2019, the country with the lowest relative housing expenditure was Malta, with 12% of overall household consumption expenditure on housing. In other countries (e.g., Finland), housing consumption constitutes over a quarter of overall household consumption expenditure (OECD, 2021). Huber (2018) measures bubbles by independent house price booms (and boom–bust cycles).

³The measurement of house price bubbles using observational data is prone to measurement errors. A further challenge is measuring the quantity and quality of housing consumption (housing services): Cross-country data on average size (square meters) and housing quality is scarce. See Piazzesi et al. (2007) for a more in-depth discussion. For example, the NIPA real housing quantities only reflect one input into the production of housing services and are often criticized for being grossly mismeasured. Similarly, Ozimek (2014) challenges the measure of housing services provided by the Bureau of Labor Statistics (the CPI weight on housing services) due to their method for measuring imputed rents. In addition, conclusions on the causal effect cannot be easily drawn because of, among others, omitted variable problems.

experimental setup and, in each period, young and old households exist. Households decide how much *housing services* and how much of all other consumption goods to consume, how much to invest in the *housing asset*, and how much to save in riskless bonds.⁴ In the subsequent period, households receive the return from their housing investment and savings. The dividend from investing in the housing asset is given by the rental income the house generates (which is determined by the current demand for housing services). As standard in the asset price literature, we define the fundamental value of a housing asset by the expected discounted flow of future dividends that the asset generates. Hence, the demand for housing consumption determines both the dividend and the fundamental value of the housing asset *endogenously*.

Our results confirm that housing bubbles are substantially and significantly larger in an economy with a weaker preference for housing services and, consequently, higher relative demand for housing services. This result holds for a wide range of bubble measures and the price-to-rent ratio. Our results survive a number of robustness checks; among them alternative explanations for housing bubbles such as liquidity constraints and differences in cash-to-asset ratios: The significantly larger bubbles in the treatment with a weak preference for housing services are driven by the endogenously resulting differences in both the dividend and fundamental value of the housing asset. We therefore can conclude that preferences and thereby demand for housing services relative to other consumption goods causally and negatively affect the size of housing bubbles and that they do so through the fundamental value.

Our study contributes to the existing literature on housing markets in three ways. First, this study highlights a novel factor that may determine the size of house price bubbles: the consumption side of housing assets.⁵ This 'consumption channel' is universally valid as long as the asset under consideration is a consumption good at the same time. Examples of other assets that entail a consumption and an investment side include artwork, classic cars, and jewelry. However, the consumption channel for bubble formation is arguably most relevant for the housing asset, given that housing consumption expenditure constitutes the largest consumption expenditure share for most economies in the world. The existing theoretical and empirical literature focuses exclusively on the investment side of housing

⁴In our context, consuming housing services when young can be best understood in terms of renting the housing asset.

⁵Studying the impact of the consumption side of the house price bubble size is different from investigating the effect of an asset on aggregate consumption. Crockett et al. (2018) and Fenig et al. (2018) show experimentally that investing in an asset allows the transfer of resources from one period to the next (consumption smoothing). In our environment, the asset itself provides consumption utility to individuals, and we investigate its impact on the size of bubbles.

and shows that intensive housing investment (e.g., measured by turnover rates) is often associated with the emergence of housing bubbles.⁶ The literature thereby establishes two key determinants for house price fluctuations; mortgage market conditions and housing supply play a key role for the magnitude of housing booms and busts.⁷

Second, this paper—one of the first laboratory experiments regarding housing markets contributes to the experimental (housing market) literature on bubble formation by pointing towards a novel channel, namely the consumption side of the asset. To our knowledge, Ikromov and Yavas (2012) and Bao and Hommes (2019) are the only experimental studies that analyze housing market features and their impact on house price bubbles. Both studies ignore the consumption side of housing, and focus on channels for house price formation that work either through the demand for housing investment or through housing supply. Ikromov and Yavas (2012) find that transaction costs and the divisibility of assets reduce the magnitude of experimental housing bubbles. Bao and Hommes (2019) show that an increase in the housing supply elasticity may stabilize speculative asset bubbles. Our paper also contributes to the vast experimental asset market literature on bubble formation. The seminal paper of Smith et al. (1988), is characterized by a finite horizon with an exogenous and decreasing FV over time. Several follow-up studies replicate and modify the experimental setting in Smith et al. (1988) to study different drivers of bubble formation. Examples include experiments that study the impact on bubble formation of: short-selling restrictions; transaction costs; and the divisibility of assets; future markets; cash availability in the economy; the nature of the fundamental value (constant versus decreasing or increasing); the impact of experience; confusion and certainty about the dividend process.⁸ These studies assume that the asset is a pure investment good, ignoring that some assets, particularly housing, have an additional consumption side.

Third, we contribute methodologically to the experimental asset market literature. Our design provides a framework in which both a market for the traded asset (the house) and a market for the dividend of that traded asset (the price of housing services) exist. In the related literature, most experiments assume an exogenous dividend for the traded asset (e.g., Marimon and Sunder (1993); Noussair et al. (2001); Ikromov and Yavas (2012)). No-

⁶The strong relationship between prices and turnover rates was first illustrated by Stein (1995). Leung (2004), Andrew and Meen (2003), Hort (2000), and Berkovec and Goodman (1996) confirmed this result. Nowadays, policy makers use turnover rates as a warning indicator for bubble formation.

⁷Local and global credit aggregates, short- and long-term interest rates matter for house price fluctuations (Claessens et al. (2009); Agnello and Schuknecht (2011); Igan and Loungani (2012)).

⁸See for example Porter and Smith (1995), Smith et al. (2000), Noussair and Tucker (2016), Noussair et al. (2001), Dufwenberg et al. (2005), Haruvy and Noussair (2006), Lei and Vesely (2009), Kirchler et al. (2012), Ikromov and Yavas (2012), or Weitzel et al. (2019). For a literature review, see chapters 29–30 of Plott and Smith (2008).

table exceptions are Jaworski and Kimbrough (2016), Weber et al. (2018), and Bao et al. (2018). Unlike these studies, in our design the dividend of the bubbly asset is determined endogenously by the demand of the *market* participants. The preferences for housing services and thus the demand for housing consumption determine the dividend of the housing asset, and hence the fundamental value of the housing asset, endogenously. The endogenous, *market*-driven dividend is a crucial and novel feature for the analysis of experimental (housing) bubbles in asset markets. Beyond this endogeneity aspect, a special feature of our model-based design is that it allows for rational bubbles. Most asset market experiments use finite horizon environments, and hence rational theoretical bubbles are ruled out by construction.⁹ Furthermore, the OLG structure allows interesting potential extensions for future research, namely, analysis of housing market interventions and changes in population dynamics.

The remainder of the paper is structured as follows: Section 2 summarizes the OLG model. Section 3 describes the design and implementation of the lab experiment as well as the treatments and the hypotheses. Section 4 explains how we measure experimental bubbles and presents the results. We also discuss potential alternative explanations for our key result and address them with corresponding robustness checks. Section 5 concludes.

2 The OLG Model

The model of Huber (2018) builds on Galí (2014), and provides an explanation of why countries with a *lower* preference for housing services experienced significantly *larger* housing bubbles during 1970–2019. We take the model framework to the laboratory and provide empirical evidence for the causal impact of the preference for housing services on the bubble size. In this section, we first illustrate the underlying mechanisms driving the theoretical result intuitively. Second, we briefly present the relevant model ingredients using simplified notation. Finally, we highlight three model predictions that we test in the laboratory.

2.1 Illustration of the Model Mechanism

Suppose that total consumption in an economy can be divided into two categories: Housing consumption, hence housing services, S, and all other types of consumption, C. Households consume housing services independent of whether they own or rent the house they live in. The larger the house and the better the quality, the more housing services a household con-

⁹Our experimental design draws upon seminal theoretical work on rational bubbles by Samuelson (1958) and Tirole (1985) and is based on an overlapping generations structure.

sumes. Further, suppose that there are two identical countries: country *High* and country *Low*. These two countries differ in one aspect only—which is the aggregate preference for housing services relative to other consumption goods. Households in country *High* have a stronger preference for housing services relative to other consumption goods. One can think of this as a country full of individuals who want to live in big and/or high-quality houses, but do not care if they can go to restaurants or the theatre very often. Hence the relative demand for housing services is strong in country *High*. This leads to a high price of housing services relative to other consumption goods in country *High*, and hence a larger consumption expenditure share is spent on housing services. In contrast, in country *Low*, households do not care that much about housing services is low. And hence the price for housing services relative to other consumption goods as well as the consumption expenditure share spent on housing services is lower in country *Low*.¹⁰

This has important implications for the FV of the housing stock in each economy. Suppose that the FV of an asset is calculated by the expected discounted stream of dividends that the asset generates. What is the dividend of a housing asset? It is the rental income that a housing asset could generate, hence the price of housing services. Thus, the FV is the expected discounted stream of the price of housing services. It follows that in country High, where the aggregate preference for housing services is stronger and the relative price for housing services is higher, the FV of the housing stock will thus be higher as well—compared to country Low.

This in turn has crucial implications for the maximum bubble size: In equilibrium, every asset needs to be affordable. It follows from the resource constraint of the economy that country *High*, where the FV of the housing stock is larger, will only have room for smaller bubbles. In country *Low*, where the FV of the housing stock is lower, there is room for larger bubbles. These predictions will be discussed more formally in Section 2.3.

2.2 Model Setup

In this exposition we focus on the household sector, as subjects play the role of households in the experiment. Without loss of generality, we simplify the notation and explain the differences to allow the reader to go back and forth between the full-fetched model and the simplified version implemented in the laboratory. The model is based on an overlapping

¹⁰Observational data shows that the aggregate consumption expenditure shares spent on housing services vary a lot across countries (e.g., OECD and Eurostat). The aggregate expenditure on housing services is calculated as the sum of actual rents (what the renters pay) and the imputed rents (the rent homeowners would pay would they rent the house they are living in).

generations structure, where a continuum of households lives for two periods (young and old). The size of each generation (young and old) is normalized to unity. In each period, a young and an old generation exist; the total population thus remains constant. Households born at time t maximize the expected utility over the lifecycle

$$u(C_{y,t}) + \xi^k v(S_t) + \gamma E_t \{ u(C_{o,t+1}) \},$$
(2.1)

where $C_{y,t}$ and $C_{o,t+1}$ denote the consumption level of the non-durable composite consumption good in period t when young and in period t + 1 when old. Consuming housing stock of size S_t yields housing service utility $v(S_t)$. The preference for housing services in country k relative to all other consumption when young $C_{y,t}$ is denoted by ξ^k . The discount factor is denoted by γ and $u(\cdot) = v(\cdot) = \log(\cdot).^{11}$

Young households receive an endowment E_t , which they allocate between consuming the bundle $C_{y,t}$, consuming housing services of size S_t , investing in housing assets of size H_t , and investing in a one-period riskless bond of value Z_t ; see Equation (2.2).¹² Investing in housing assets is equivalent to purchasing housing stock of size H_t . The return on bond investments, Z_t , is given by the interest rate R_t .

The dual motives of housing are disentangled by modeling the consumption aspect (consuming housing services S_t) and the investment aspect (investing in housing assets H_t) separately. This assumption distinguishes this model from existing models of rational housing bubbles and allows to isolate the impact of the demand for housing services on the house price bubble size. This model provides the simplest framework to investigate the research question—does the demand for housing consumption drive the size of housing bubbles? All young households consume housing services S_t by renting housing stock from the old generation, and invest in housing by buying housing stock H_t from the old generation. We thus do not distinguish explicitly between renters and homeowners.¹³ Given

¹¹Housing services and *all other* composite consumption goods are assumed to be separable (as e.g., in Iacoviello (2005)). The log specification over S and C is based on findings of Davis and Ortalo-Magne (2011), who show that the expenditure share on housing services is constant (over time and across US cities). Also, Huber (2018) finds for a sample of 18 OECD countries that cross-country differences in consumption expenditure shares on housing services are constant over time.

¹²The full-fledged model of Huber (2018) follows Galí (2014), where firms produce the composite consumption good under the assumption that *young* households supply their labor services inelastically for a wage W_t . In both models, the "financial wealth" of the *young* consists of two elements: the wage W_t and an endowment U_t (that follows an exogenous i.i.d stochastic process), hence $E_t = W_t + U_t$. In the experiment we are interested in comparative statics of the deterministic steady state, hence the model boils down to an endowment economy, where households receive a constant endowment E = W + U at the beginning of the lifecycle.

¹³Theoretically, the model allows for the possibility that some households consume more housing than they own S > H (i.e, they would be renters and landlords at the same time—renting a larger property to live in while investing/renting out a smaller housing asset). Or vice-versa, they may

our research question, such a differentiation is not of interest and also in the real–world households consume housing services independently of whether they own or rent the house they are living in.¹⁴

Investing in the housing asset H_t yields a dividend payment in the subsequent period a rental income when *old*. Before *old* households die, they sell their housing stock to the new young generation. The budget constraint of the young household at time t is given by

$$P_t C_{y,t} + P_t^r S_t + Q_t H_t + Z_t \le E_t, \tag{2.2}$$

where P_t denotes the price of the composite consumption good and P_t^r the price for housing services. The purchasing price of one unit of housing stock is denoted by Q_t , and is defined as a sum of the fundamental part FV_t and a bubble component B_t :

$$Q_t \equiv FV_t + B_t, \tag{2.3}$$

where the fundamental part is defined as the expected discounted stream of dividends (rental income) the housing asset H_t generates:

$$FV_t \equiv E_t \left\{ \sum_{j=1}^{\infty} \left(\frac{1-\delta}{R_{t+j}} \right)^j P_{t+j}^r \right\}.$$
 (2.4)

The budget constraint when *old* is given by Equation (2.5). By purchasing the consumption bundle $C_{o,t+1}$, the *old* household consumes all its financial wealth. The *old* household's wealth consists of (1) the rental income from renting its housing stock to the *young* generation $P_{t+1}^r H_t$, (2) the reselling value of its housing stock $Q_{t+1}H_t$, and (3) the payoff from its maturing bond holding $R_t Z_t$. Formally, for each *old* household we have

$$P_{t+1}C_{o,t+1} \le (1-\delta) \left(P_{t+1}^r + Q_{t+1} \right) H_t + R_t Z_t.$$
(2.5)

Equation (2.5) shows that the value of the housing asset depreciates by the fraction $\delta \in [0, 1)$ at the end of each period. However, this depreciation is compensated by the endowment of the *young* such that the total value of the housing stock in the economy remains constant.¹⁵ To implement the original model in the laboratory, we make one

consume less housing than they own S < H (i.e., they would be homeowners and landlords—living in and hence consuming housing of size S and being landlord of the remaining fraction H - S).

¹⁴The motivation to not distinguish between renting and owning goes back to Henderson and Ioannides (1983): "[...]before the introduction of institutional considerations there is no reason for people to actually owner-occupy their consumption–investment demands, as opposed to being landlords of their asset holdings and renting their consumption demand from some other landlords".

¹⁵When born in t, young households receive an endowment E_t . A part of this endowment is

additional assumption: the dividend payment that the *old* household receives is also subject to depreciation. Note that this modification does not change the model's qualitative predictions. However, this assumption leads to a simplified FV_t calculation as shown in (2.4)—which is equal to the implemented FV_t calculation in the experiment where we have an indefinite horizon by construction. In the experiment, we assume a constant continuation probability (1-x) and no depreciation of the housing which leads to an identical FV_t calculation; i.e., $FV_t \equiv E_t \left\{ \sum_{j=1}^{\infty} \left(\frac{1-x}{R_{t+j}} \right)^j P_{t+j}^r \right\}$.¹⁶

2.3 Model Predictions

This section presents the model predictions that result from comparative statics of the steady state when varying the model parameter of interest ξ . We test these model predictions in the laboratory experiment. Comparative statics of the steady state show that an increase in ξ (capturing a rise in households' preference for housing services relative to other consumption goods) induces an *intra*-temporal reallocation: a substitution from other consumption goods C_y toward housing services S when *young*. This leads to an increase in the consumption expenditure share spent on housing services S and to a decrease in the consumption expenditure share spent on other consumption goods C_y when *young*¹⁷:

Share_S =
$$\frac{\text{spending on housing services S}}{\text{total spending on consumption}}$$

= $\frac{P_t^r S_t}{P_t^r S_t + P_t C_{y,t}} = \frac{\xi}{1+\xi},$ (2.6)

This change in the consumption composition (i.e., consumption expenditure shares) is also reflected in the relative price $p_t^r = P_t^r/P_t$, the price for housing services relative to other consumption goods. The increase in the demand for housing services S_t leads to a higher absolute P_t^r and relative price p_t^r . It follows that the fundamental value FV_t , the expected discounted stream of the price for housing services, increases.

Prediction 1: The Absolute Bubble Size B_t decreases in ξ . In equilibrium it must hold

equal to the value of δ units of housing stock, whose price is $Q_t > 0$. Formally, $U_t = \delta Q_t H_t$.

¹⁶Hence, the actual discount factor $[(1-x)/R_{t+j}]^j$ is identical with $\delta = x$. Note that the assumption of a zero depreciation rate in the experiment comes without loss of generality, but simplifies the information subjects need to process. With a positive depreciation rate, the actual discount factor in the experiment would be given by $[(1-\phi)/R_{t+j}]^j$, where $\phi \equiv \delta + x$.

discount factor in the experiment would be given by $[(1 - \phi)/R_{t+j}]^j$, where $\phi \equiv \delta + x$. ¹⁷Share_C = $\frac{\text{spending on C}}{\text{total spending on consumption}} = \frac{1}{1+\xi}$; note that these shares are equal to CPI weights. The official CPI weight on housing services of a country includes both, the spending on housing services of renters (the actual rent they pay) and the spending on housing services of homeowners (the imputed rent, what they would pay if they would rent the house they are living in).

that

$$B_t \in [0, \ E - FV_t] \quad \forall \quad t. \tag{2.7}$$

Equation (2.7) states the condition for bubble existence and the theoretical upper bound for the housing bubble defined by overpricing of the housing stock $Q_t - FV_t$. This condition is derived from the budget constraint of the *young*, Equation (2.2). The larger the preference for housing services, the larger the fundamental value FV_t of the housing stock today ($\partial FV_t/\partial \xi > 0$). It follows from Equation (2.7), that the theoretical maximum of the aggregate bubble size is smaller; i.e. ($\partial B_t/\partial \xi < 0$). In other words, countries that are characterized by a larger fundamental value of real estate (potentially because the demand for housing consumption is stronger) have less room for large bubbles—that is to say, cannot experience large bubbles—compared to countries where the fundamental value of real estate is lower (i.e., the demand for housing consumption is weaker).

Prediction 2: The Relative Bubble Size B_t/FV_t decreases in ξ . The model predicts that the realized relative bubble size B_t/FV_t , that is the overpricing $Q_t - FV_t$ divided by the fundamental value FV_t , is larger the weaker the preference for housing services relative to other consumption goods ξ , i.e. $(\partial (B_t/FV_t)/\partial \xi < 0)$.

Prediction 3: The Price-to-Rent Ratio PRR_t decreases in ξ . In reality, it is challenging to detect a housing bubble $B_t = Q_t - FV_t$ given the difficulty of measuring the fundamental value FV_t of a housing asset correctly. In policy and public debates, the price-to-rent ratio is often referred to as an indicator to detect whether housing markets are fairly valued, or in a "bubble." This indicator is provided to the public on a country-level, but also for large cities. For example the UBS Global Real Estate Bubble Index uses this metric to determine which cities are at the greatest risk of a bubble—the higher the PRR, the greater the risk. In the model, the PRR is given by Q_t/P_t^r and decreases in ξ , i.e. $(\partial PRR_t/\partial\xi < 0)$.

3 Experimental Design

This section describes how we take the OLG model to the laboratory. First, we present the treatments. Second, we explain the decisions that subjects, who assume the role of households, make in a lifecycle. Third, we describe the assignment to markets and the overall structure of the experiment. Finally, we present the procedure and the subject pool. The instructions—including the decision screens—are provided in Appendix D.

3.1 Treatment Variation and Calibration

In order to test the aforementioned predictions, we implement the following two treatments in a between-subject design (subjects are randomly assigned to either treatment):

Treatment $(\xi = 2)$. Households have weak preferences for housing services (low ξ). The utility from consuming housing services S relative to the consumption good C is low.

Treatment $(\xi = 6)$. Households have strong preferences for housing services (high ξ). The utility from consuming housing services S relative to the consumption good C is high.

In the model, the preference parameter ξ determines the consumption expenditure spent on housing services as a fraction of the total consumption expenditure in equilibrium, see Equation (2.6). Hence, we can calibrate the parameter ξ using observational data for the housing services expenditure shares. These shares vary from 12% to 30% (sample of 18 OECD countries).¹⁸ The implied parameter ξ is roughly three times larger for countries with the highest expenditure share compared to countries with the lowest expenditure share. In our experiment, we match this relative difference in magnitude of parameter ξ . In the treatment ($\xi = 6$), the preference parameter ξ is three times larger than in the treatment ($\xi = 2$). Appendix Table A1 summarizes the remaining parameter choices.

3.2 Decisions in a Lifecycle (Young and Old)

The seminal OLG laboratory experiment is that of Marimon and Sunder (1993)—who address questions of equilibrium selection and sunspots in the presence of multiple equilibria.¹⁹ In our overlapping generation design, a household's lifecycle consists of two periods (*young* and *old*) and each subject in the experiment assumes the role of a household: In the first period, subjects make decisions as a *young* household. In the second period, subjects make decisions as an *old* household and receive payments that are based on their decisions when *young* and *old*, as well as aggregate outcomes. Figure 1 shows the lifecycle and the transactions that take place when *young* and *old*.

¹⁸Rewriting Equation (2.6) gives $\xi = Share_S/(1 - Share_S)$, the parameter ξ is uniquely pinned down for each country. Appendix Table shows the observational country data for *Share_S*.

¹⁹Our experimental design differs in three aspects. First and as Camera et al. (2003), we use a continuous double auction design. Second, we construct 'generations' by randomly assigning subjects to one of two markets after each life cycle instead of having them wait to be reassigned to participate in the one market. Third, only one randomly chosen lifecycle forms the basis of payments. To implement an infinite time horizon in the laboratory, we follow Crockett et al. (2018) and implement an indefinite horizon. These features are discussed in detail in this section. Other OLG laboratory experiments include: Lim et al. (1994) to study price dynamics and the use of money as a store of value, Bernasconi and Kirchkamp (2000) to investigate how inflation is determined by monetary policy and average savings, and Camera et al. (2003) to study characteristics of economies that encourage the use of fiat money (a dominated asset).



Figure 1: Lifecycle and Transactions when Young and Old.

At the beginning of a lifecycle, each household receives an endowment E_t . In accordance with the budget constraint in Equation 2.2, households can spend the endowment E_t by consuming the composite consumption good $C_{y,t}$ and housing services S_t , and by investing in the housing asset H_t and in the riskless bond Z_t .

Young households decide how many units of the consumption good $C_{y,t}$ and how many units of housing services S_t they want to purchase. Following a similar design as in Fenig and Petersen (2017) and Fenig et al. (2018), subjects do so by clicking on a combination of $C_{y,t}$ and S_t in a graph on the decision screen. Young subjects get utility as defined by the first two elements in Equation (2.1). The utility level for all possible combinations of $C_{y,t}$ and S_t is shown in a colored heat map and, for the selected combination, the number is displayed next to the heat map. The preference for housing service relative to other consumption goods (denoted by the parameter ξ) affects the utility a household gets from consuming a certain combination of $C_{y,t}$ and S_t .

Our treatment variations exogenously induce varying preferences for housing services. As explained in Section 3.1, preferences for housing services S_t relative to other consumption goods $C_{y,t}$ are either low ($\xi = 2$) or high ($\xi = 6$), depending on the random treatment assignment. It is crucial that subjects understand the consequences of their consumption decisions for their household utility. The heat maps visualize the treatment variations and the displayed numeric utility further enhances subjects' comprehension (see screenshots in Appendix D).

The general equilibrium model entails a market for housing consumption S_t and one for housing investment H_t . The price for housing consumption P_t^r defines the dividend of the current *old* households' housing asset holdings H_t . The dividend of the housing asset H_t is thus endogenous. In most experimental asset markets, the dividend process is exogenous and common knowledge. Our experimental design's critical and novel feature is the endogenous nature of the dividend of the housing asset H_t . Our design thereby contributes to the nascent but notable literature on endogenous dividends.²⁰ In our setup, households' demand for housing services relative to other consumption goods determines the dividend. Hence, the price for one unit of S_t , $p_t^r = P_t^r/P_t$, depends on all *young* household's purchases of $C_{y,t}$ and S_t in the market. The price for one unit of the consumption good $C_{y,t}$, P_t , is set to the numeraire (and equal to one).

The market price, P_t^r , can only be determined once all young households in the market have submitted their purchase decisions for $C_{y,t}$ and S_t (subject to the budget constraint in Equation (2.2) and the exogenously available supply; i.e. C^{supply} and S^{supply}). We therefore provide a graphical tool on which subjects can simulate the purchase decision with regard to $C_{y,t}$ and S_t of other young households in the market. Together with the own chosen combination of $C_{y,t}$ and S_t , the price for one unit of S_t , P_t^r , is calculated and displayed on the screen.²¹ We ask young households to submit the maximum number of units of $C_{y,t}$ and S_t they want to purchase. The price P_t^r is fully flexible and can take values between 0 and infinity. The exact calculation of P_t^r is shown in Appendix Table A1.²² Once all young households in the market have submitted their demands, the resulting market price for one unit of S_t , P_t^r , as well as the finally purchased units of the consumption good $C_{y,t}$ and housing service S_t are displayed on the screen.

Young households also purchase units of housing asset H_t in a continuous double auction (for instance, Camera et al. (2003)) from the current old in the market. Before the young households get to the double auction, they learn how the dividend of the housing asset H_t will be determined—that is, that it will depend on the choices with regard to $C_{y,t+1}$ and S_{t+1} by the future young (the generation born in t + 1). Young subjects can simulate the average purchase of $C_{y,t+1}$ and S_{t+1} by the future young using a graphical tool.²³ Remember that the dividend is determined endogenously by the rental price in the

 23 As a help device, the same heat map is depicted because the *future young* will make the exact

²⁰Weber et al. (2018) study assets with endogenous, price-dependent default probabilities. In contrast to our study, the asset has no consumption side, and the dividend payment (the interest rate) is fixed and known. However, the default probability of the asset's issuer is endogenous, and hence the realized dividend (and FV) is endogenous. In Bao et al. (2018) and Jaworski and Kimbrough (2016), the dividend is endogenously determined by the CEO's effort and the monopolist's decision, respectively. In the latter work, the dividend is also determined by selling a good to demand–revealing robot buyers.

²¹Subjects can try as many combinations as they wish (without time restriction). This step helps subjects learn how aggregate decisions affect prices and dividends in the economy and their payoffs. Once they decide on a combination $(C_{y,t}, S_t)$, they click the "Submit" button.

²²Alternative procedures to determine the market demand and market price for housing services S_t include the elicitation of full demand schedules or call markets. In our setup, *young* households demand *two* consumption goods simultaneously, which complicates their implementation substantially. In Appendix A, we provide evidence that our setup allowed subjects to form adequate point estimates about P_t^r .

subsequent period, P_{t+1}^r . To help households form expectations about the rental return of housing asset H_t , the dividend from each simulated combination of $C_{y,t+1}$ and S_{t+1} is calculated and displayed on the screen. We implement a standard experimental double auction with the only exception that *young* households can only *purchase* and *old* households can only *sell* housing assets H_t . Young subjects can initiate a purchase of an asset by submitting an offer to buy (a price for which they want to buy one unit of housing asset H_t) or by accepting an offer to sell submitted by *old* households (a price for one unit of housing asset H_t). The duration of the double auction is three minutes.²⁴ After the double auction is over, the remaining budget remains automatically in the bank account (= is invested into a riskless bond Z_t) and earns a fixed interest payment of 5%.

When old, households learn about their investment return from asset H_t —that is, they receive a dividend payment for each housing asset H_t they bought when young.²⁵ Subsequently, the old households enter a double auction in which they can sell housing assets H_t to the current young households in the market. Old households can initiate a sale of an asset by submitting an offer to sell (a price for which they want to sell one unit of housing asset H_t) or by accepting an offer to buy from the young households (a price for one unit of housing asset H_t). The dividend from the sold housing assets H_t , the sales price of housing assets H_t , and the return from the riskless bond Z_t are automatically spent on consumption good $C_{o,t+1}$. The utility from consuming $C_{o,t+1}$ is defined by the third element of the utility function in Equation (2.1).

To facilitate decisions and ensure that decisions are as well-informed as possible, subjects play four practice periods before the actual experiment starts and can access a history screen from any decision or feedback screen (and go back to the decision or feedback screen).²⁶ At the end of a lifecycle—subjects receive summary information on their decisions in the corresponding lifecycle: the number of units $C_{y,t}$ and S_t purchased when young and the respective prices, P_t and P_t^r , as well as the number of units of H_t purchased. Subjects receive information on the dividend of each purchased housing asset H_t , the price for

same decision on purchasing $C_{y,t+1}$ and S_{t+1} as the current young households in period t.

²⁴There is no depreciation of the housing stock; thus the stock of housing assets H_t remains constant from period to period. Therefore, unsold units of the housing asset H_t are assigned randomly to the current *young* households in the market at a punishment price. The punishment price is 50% less (more) than the median trading price for the current *old* (*young*) in the market. This incentivizes subjects to trade the existing housing stock H_t , such that the market clears. On average, less than one unit of housing stock H_t remains unsold per experimental market and period (no significant treatment differences) suggesting that it does not affect the experimental results.

²⁵The dividend of the housing asset H_t is given by the price for housing services P_{t+1}^r , which is determined by the demand for C_{t+1} and S_{t+1} of the current young households in that period.

²⁶The history screen shows for all previous periods a summary of (1) the subject's decisions on C_t , S_t , H_t ; (2) the corresponding utilities; (3) the return from investing in the riskless bond Z_t ; (4) the median price for all traded housing assets; and (5) the average dividend per housing asset.

which they sold the purchased assets H_t , the median price of all sold housing assets, Q_t , the return from the riskless bond Z_t , and the total lifecycle utility (Equation (2.1)).

3.3 Market Assignment and Experimental Procedure

In the model, infinitely many households exist, which—for obvious reasons—is not feasible in the laboratory. Therefore subjects, who represent households, play several lifecycles as in Marimon and Sunder (1993). Each session is composed of 16 subjects and two markets A and B. At the beginning of the experiment, 50% of all subjects are randomly assigned to Cohort I and the remaining 50% of subjects to Cohort II (eight in Cohort I and eight in Cohort II). All subjects are informed that they will remain in the assigned cohort for all periods of the experiment.

At the beginning of period 1, four members of each cohort are randomly assigned to market A and the other four members to market B. Cohort I (II) starts as *young* (*old*) households in period 1 and subjects make decisions accordingly. Appendix Figure A1 presents an overview over each cohort's lifecycles. From period 2 onwards, cohorts switch between generations in each period. That means that Cohort I (II) takes the role of *old* (*young*) households in period 2 and takes the role of *young* (*old*) households in period 3, etc. Cohort I's lifecycle 1 consists of periods 1 and 2; its lifecycle 2 consists of periods 3 and 4, etc. Cohort II's lifecycle 1 consists of periods 2 and 3; its lifecycle 2 consists of periods 4 and 5, etc.

Our incentive structure differs from existing OLG asset market experiments. Only one completed lifecycle (chosen randomly) forms the basis for the payment. We pay only one completed lifecycle because this most closely aligns incentives with the infinite-horizon OLG model where the death probability when young is zero and households only live for one lifecycle. This design choice prevents subjects from hedging risk by playing different strategies in different lifecycles.

As an important design feature, subjects are randomly assigned to either market A or market B at the beginning of a lifecycle and remain in the same market throughout a lifecycle. That means that the formerly *young* households remain in the *same* market when turning *old*, while the formerly *old* households are *randomly* assigned to either market A or market B before a new lifecycle starts. Subjects know in which market they are (A or B), but do not know the other market participants' identity. In comparison to the existing literature on OLG market experiments,²⁷ the way we assign subjects to markets keeps the

²⁷In Marimon and Sunder (1993)'s design, each subject plays during two periods (i.e., a lifecycle) as young and old in the first and second period, respectively. After the second period, (old) subjects

design close to the OLG model and has the advantages of gathering more observations, providing the subjects with experience, reducing repeated interaction with the same subjects and collusive behavior in small markets, as well as simplifying the complex setup. The instruction handouts illustrate the assignment to cohorts and markets (Appendix D).

To implement an infinite time horizon in the lab, we follow Crockett et al. (2018) and implement an indefinite horizon by assuming a constant probability of continuation in each period. As Duffy (2016) points out, "infinite horizons are not operational in the lab, but indefinite horizons are". The probability of continuation is set to 80%. As discussed in Section 2.2, the stopping probability captures for instance the depreciation rate in the theoretical model.²⁸ Before running the sessions, we threw a ten-sided dice repeatedly until the number 0 or 1 turned up to determine the number of periods in the experimental session. Thus, the length of each session is the same and equal to nine periods. Both the theoretical and the actual length of a session remains the same across treatments. Before the experiment starts, subjects have four trial periods to get familiar with the experimental environment.

3.4 Procedure and Subject Pool

At the beginning of each experimental session, the instructions, illustrating screenshots, graphs, and tables, are handed out to the subjects on paper and read aloud by one experimenter. The beginning of the trial periods and the start of the incentivized periods are also announced aloud. Appendix D shows the material handed out to the subjects. The instructions and materials are identical for treatment ($\xi = 2$) and treatment ($\xi = 6$) except for one variation: The formula for the lifetime utility differs depending on the treatment: $u(C_{y,t})+2*v(S_t)+E_t\{u(C_{o,t+1})\}$ for treatment ($\xi = 2$) and $u(C_{y,t})+6*v(S_t)+E_t\{u(C_{o,t+1})\}$ for treatment ($\xi = 6$). The heat maps—illustrating households' utility from consumption goods $C_{y,t}$ and housing services S_t —and screenshots are adjusted accordingly.

The sessions were conducted at the BEElab at Maastricht University in April, May, and September 2016, and the programming was done with the software z-Tree (Fischbacher, 2007). Participants were mainly Maastricht University undergraduate students, recruited using the online recruitment system ORSEE (Greiner et al., 2004). We sent invitations to students from the following study fields: Econometrics and Operations Research, Economics and Business Economics, Fiscal Economics, and International Business.

are randomly assigned to restart as young participants or to wait until being reassigned.

 $^{^{28}}$ Duffy (2016) argues that a continuation probability has the advantage that future payoffs are discounted at a certain rate (discount factor with infinite horizon) and the stationarity of an infinite horizon is induced.

In total 256 subjects took part in 16 experimental sessions (eight sessions per treatment) composed of 53% women and 47% men (the share of women per session varied between 37.5% and 62.5%).²⁹ The average age was 21 years. The conversion rate was 1 utility unit to 3 euros, and the average earnings per subject were 27.27 euros (SD = 4.63 euros).³⁰ This includes a show-up fee of 5 euros, a finishing fee of 5 euros, and the converted utility units of one completed lifecycle that was randomly selected at the end of the experiment. The average duration of a session was 2.5 hours. After the experiment had finished, subjects were asked to fill out a questionnaire and were paid their earnings in private.

4 Data Analysis and Results

This section presents the test results of the model predictions in Section 2.3. Section 4.1 explains how we measure experimental house price bubbles. Section 4.2 presents the descriptive statistics and the experimental results on the impact of housing consumption on the bubble size. Section 4.3 shows that the bubble size is constant over time as predicted by the OLG model. Sections 4.4 and 4.5 describe robustness checks that rule out two alternative explanations for the treatment effect, namely liquidity constraints and (small endogenous) differences in cash-to-asset ratios. We refer to Appendix C for additional robustness checks (e.g., on expectation formation, alternative bubble measures and trading prices (median instead of mean), reaction to the treatment variation).

4.1 Measuring the Experimental Bubble Size

To measure the housing bubble in the experiment, we need to compute the fundamental value—that is, the *expected* discounted stream of dividends. We thus need to take a stance on subjects' expectation formation about the future stream of dividends in the experiment. We could do so by—consistent with the OLG model—assuming that households are rational as in Equation (2.4). Given that the OLG model has no information frictions, the agents will forecast the future stream of dividends correctly in the deterministic steady state. In the steady state of the model, where P^{r*} denotes the steady state value of the dividend, Equation (2.4) simplifies to $FV_t = \frac{(1-\delta)}{R-(1-\delta)}P^{r*}$ $\forall t$. Using the same calibration of parameters as in the experiment ($R = 1.05, \delta = 0.2, S = H = 20$) gives a FV of 19.2

²⁹Eckel and Füllbrunn (2015) and Holt et al. (2017) show that asset markets with a declining dividend and a higher share of male participants produce larger bubbles. Though Holt et al. (2017) show no gender differences in markets with a constant dividend, we invited the same number of male and female students to each session.

³⁰The average earnings were 23.12 euros (SD = 1.81 euros) in treatment ($\xi = 2$) and 31.42 euros (SD = 2.25 euros) in treatment ($\xi = 6$).

for the treatment ($\xi = 6$), and a FV of 6.4 for the treatment ($\xi = 2$). All the results that follow, are robust to assuming the model–consistent rational expectation formation.³¹

The empirical literature on expectation formation has however shown that households are to a large degree backward looking; that is, they adapt their expectations to past observations. Adaptive expectations seem to match survey and experimental data rather well.³² We therefore assume that subjects recognize all realized dividends of the past and update their beliefs accordingly.³³ Hence, in period one, we assume that households expect all future dividends to be equal to the current and first realization. In all future periods, households update their belief and expect that all future dividends will be equal to the average of all realized dividends to date. Hence, households expect the following fundamental value:

$$FV_t \equiv E_t \left\{ \sum_{j=1}^{\infty} \left(\frac{1-x}{R} \right)^j P_{t+j}^r \right\},\tag{4.1}$$

where (1-x) denotes the common-knowledge continuation probability, R the interest rate, and $E_t \left\{ P_{t+j}^r \right\} \equiv \sum_{k=1}^t \frac{P_k^r}{t} \equiv \bar{P}_t^r$ the average of the realized dividends in period t.

Having defined the expectation formation and hence the fundamental value FV as in Equation (4.1), we can compute the bubble measures. We use both theoretical and experimental bubble measures to analyze the treatment differences. The theoretical bubble measures are derived from the OLG model in section 2: the absolute bubble size $B_t = Q_t - FV_t$, where Q_t denotes the realized trading price in the experiment in period t(prediction 1), the relative bubble size B_t/FV_t (prediction 2), and the price-to-rent ratio $PRR_t = Q_t/P_t^r$ (prediction 3). In the experimental asset price literature, there seems to be a converging consensus on the suitability of the bubble measures proposed by Stöckl et al. (2010)—the relative absolute deviation RAD and the relative deviation RD. We use these two experimental indicators for measuring the bubble size. Doing so facilitates the comparison of our results with the results (and bubble sizes) of other asset market experiments.

The Relative Absolute Deviation, $RAD = \frac{1}{N} \sum_{t=1}^{N} \frac{|Q_t - FV_t|}{|FV|}$, measures the average level of "mispricing". N denotes the total number of realized periods in the experiment and \bar{FV} the absolute value of the average fundamental value of the market, with $\bar{FV} = \frac{1}{N} \sum_{t=1}^{N} FV_t$.

³¹Results are available upon request.

³²Hommes (2020)'s survey of the (behavioral macro and experimental) literature shows that individuals use simple, backward-looking expectation rules such as adaptive expectations or trendfollowing rules. Specifically in the context of housing markets, the recent empirical literature on expectation formation shows that households update their expectations by extrapolating past information about aggregate variables (e.g., Armona et al. (2019)).

³³Appendix C presents two alternative types of adaptive expectation formation. Our results are robust to these variations.

The Relative Deviation, $RD = \frac{1}{N} \sum_{t=1}^{N} \frac{(Q_t - FV_t)}{|FV|}$, measures "overvaluation". While RAD averages the absolute difference between the mean price and the fundamental value, RD averages the difference between the mean price and the fundamental value.³⁴

4.2 Impact of Preference for Housing Services on Bubble Size

This section presents our main experimental results comparing the bubble size in treatment ($\xi = 2$) and treatment ($\xi = 6$). Recall that we use three bubble-magnitude measures derived from the theoretical model ($B_t = Q_t - FV_t$, B_t/FV_t , and PRR_t) and two experimental bubble-magnitude measures (RAD and RD). Figures 2–4 show the time series of the respective three theoretical bubble-magnitude measures disaggregated by treatment.³⁵ Table 1 shows the results of non-parametric tests and regression analysis of treatment differences; thus, testing the model predictions in Section 2.3. Here, the results for the three theoretical and the two experimental bubble-size measures are depicted.

To determine whether treatment differences are statistically significant, we conduct pairwise Mann–Whitney U tests for each bubble size indicator. The first two rows of Panel A in Table 1 show the average value of the five (theoretical and experimental) measures of bubble magnitude for each treatment.³⁶ The last row of Panel A presents the corresponding Z-value and significance level. We consider conservatively each session as an independent observation because, within one session, subjects go back and forth between the two markets A and B. Session–level measures are constructed by taking the average of the measures of the two simultaneous markets resulting in eight independent observations per treatment. Panel B in Table 1 Panel B reports Ordinary Least Square regression results on treatment differences for all five bubble measures. The dependent variable are the five bubble–size measures, respectively, and the explanatory variable of interest is the treatment dummy ξ that is equal to one for the treatment ($\xi = 2$), and zero otherwise. The standard errors are (conservatively) clustered at the session–level. We control for the gender composition of the session and for the period of the experiment to account for potential time trends that could result from learning by experience.³⁷

³⁴Appendix C shows that our results are robust to using three alternative bubble indicators: the Price Amplitude (PA), first proposed by King et al. (1991); the Total Dispersion (TD) and the Average Bias (AB), both first introduced by Haruvy and Noussair (2006). Also, the results are robust to using the median trading price Q_t^m instead of the mean Q_t to compute the bubble sizes. ³⁵Appendix Figures B1 and B2 show the time series of the fundamental value FV_t and the mean

trading price Q_t for each session and market separately.

³⁶Appendix Table B1 provides descriptive statistics at *a session level* for all five bubble measures. ³⁷We use periodic observations for the three theoretical measures and session–level observations for the two experimental measures. The results are robust to using the averages of each session (N = 16). None of the controls is statistically significant; the results are robust to their exclusion.

Result 1 (The Absolute Housing Bubble Size decreases in ξ). Households' preference for housing services affects the absolute bubble size B_t negatively: B_t is significantly smaller in treatment ($\xi = 6$) than in treatment ($\xi = 2$).

The first model prediction states that the absolute bubble size B_t —the difference between the trading price Q_t and the fundamental value FV_t —is smaller, the stronger the preference for housing services (relative to other consumption goods). Figure 2 provides the corresponding comparative statics and shows the average mean trading housing price Q_t (solid line with squares) and its fundamental value FV_t (dashed line with circles) for each period and for both treatments. The left panel shows the averages over sessions in treatment ($\xi = 2$), the right panel the averages over sessions in treatment ($\xi = 6$). Inducing a strong preference for housing services (relative to other consumption goods) leads to a statistically significantly higher FV (18.37) compared to the treatment with a weak preference for housing services (6.31) (z = -3.363, p = 0.0002). These values closely match the model's predicted steady state values (19.2 and 6.4), according to Equation (2.4).

Figure 2 shows that the housing asset is overvalued in both treatments and illustrates our key result: The absolute bubble size B_t is much smaller in the right panel—when the preference for housing services is strong. To highlight this key result, we add two point-dotted lines to Figure 2 indicating the range $[(1 - 60\%) \cdot FV_t, (1 + 60\%) \cdot FV_t]$. We choose this particular range because for treatment ($\xi = 6$), the trading price Q_t just lies in between these bounds, whereas for treatment ($\xi = 2$), the trading price Q_t is far outside of this range.³⁸ The first column of Table 1 reports that the treatment difference in the absolute bubble size B_t (22.03 versus 10.61) is statistically significant at the 1% level.

Figure 2 reveals that the fundamental value FV_t and the average mean trading price Q_t , and thus the absolute bubble size B_t , are relatively constant over time in both treatments.³⁹ Similarly, Figure 3 illustrates that the relative bubble size B_t/FV_t is relatively stable over time in both treatments. Recall that we implemented two deterministic steady states of the model in the lab, one with the key model parameter $\xi = 2$ and one with $\xi = 6$. In each deterministic steady state, the absolute and relative bubble size are constant and larger in the steady state with a weaker preference for housing services ($\xi = 2$). And this is exactly

³⁸Figure 2 reveals that the price Q_t is similar across treatments. Q_t^{max} denotes the maximally feasible average trading price that would prevail if participants would invest—after the consumption decisions (S_t and C_t)—all their remaining budget into the housing asset (i.e., no saving via riskless bond). Figure 2 shows that Q_t is much smaller than Q_t^{max} in both treatments.

³⁹The roughly constant trading price Q_t may trigger concerns that the participants did not react to the treatment differences. This is not the case as our robustness check 6 in Appendix C shows. The slight decrease in the trading price over time in treatment ($\xi = 2$) is driven by one outlier, session 12 (Appendix Figure B1).

what we find. This paper provides a thorough discussion of "flat experimental bubbles" and on how our results contribute to this related literature by Hoshihata et al. (2017) and Kopanyi-Peuker and Weber (2021) in Appendix B.



Figure 2: Average Mean Trading Price Q and FV (by Treatment). Notes: ξ denotes the preference for housing services S_t (relative to consumption goods C_t).

Result 2 (The Relative Housing Bubble Size decreases in ξ). Households' preference for housing services affects the relative bubble size B_t/FV_t negatively: B_t/FV_t is significantly smaller in treatment ($\xi = 6$) than in treatment ($\xi = 2$).

The second model prediction states that the relative bubble size is smaller the stronger the preference for housings services (relative to other consumption goods). For each session, Figure 3 shows the time series of the bubble relative to the fundamental value B_t/FV_t . Comparing the left ($\xi = 2$) and right ($\xi = 6$) panels shows that the relative bubble size is substantially smaller when the preference for housing services is strong. The second column of Table 1 shows that this treatment difference in the average relative bubble size B_t/FV_t (3.50 versus 0.58) is statistically significant (p < 0.01).

Result 3 (The Price-to-Rent Ratio decreases in ξ). Households' preference for housing services affects the Price-to-Rent Ratio (PRR) Q_t/P_t^r negatively: Q_t/P_t^r is significantly smaller in treatment ($\xi = 6$) than in treatment ($\xi = 2$).

The third prediction, derived from the OLG model, concerns the Price-to-Rent Ratio. Comparative statics of the deterministic OLG model show that the stronger the preference for housings services (relative to other consumption goods), the smaller the PRR. Figure 4 plots the Price-to-Rent Ratio at a session-level for each treatment and shows that the PRR is consistently smaller in the treatment ($\xi = 6$) compared to the treatment ($\xi = 2$). The third column of Table 1 reports that this treatment difference in the average Priceto-Rent Ratio (14.28 versus 5.07) is also statistically significant at the 1% level.



Figure 3: Relative Bubble Size B_t/FV_t (by Treatment and Session). Notes: ξ denotes the preference for housing services S_t (relative to consumption goods C_t).



Figure 4: Price-to-Rent Ratio PRR_t (by Treatment and Session). Notes: ξ denotes the preference for housing services S_t (relative to consumption goods C_t).

To facilitate a comparison of our results with other asset market experiments, we use two well-established bubble indicators in the experimental asset price literature (RAD and RD). Table 1 shows the average values of both indicators in the last two columns and compares the distribution of values across treatments. For the treatment ($\xi = 2$), RAD indicates that the periodic trading price is on average 349% larger than the FV in the market. This compares to 65% in the treatment ($\xi = 6$). According to the RD indicator, the average trading price per period is 349% above the average FV when the preference for housing services is weak. This compares to 58% when the preference for housing services is strong—a large difference. Table 1 confirms that these treatment differences are significant for both indicators (p < 0.01).

Result 4 (RAD and RD decreases in ξ). Households' preference for housing services affects the experimental bubble measures (RAD and RD) negatively: RAD and RD are significantly smaller in treatment ($\xi = 6$) than in treatment ($\xi = 2$).

	B_t	B_t/FV_t	PRR_t	RAD	RD
Panel A: MWU tests					
Treatment $(\xi = 2)$	22.03	3.50	14.28	3.493	3.493
Treatment $(\xi = 6)$	10.61	0.582	5.070	0.651	0.581
Z	2.626^{***}	3.361^{***}	3.361^{***}	2.731^{***}	3.361^{***}
Panel B: OLS regressions					
Treatment dummy $(\xi = 6)$	-11.04***	-2.929***	-9.253***	-2.873***	-2.926***
	(3.663)	(0.409)	(1.280)	(0.422)	(0.434)
Constant	31.79^{*}	3.431^{*}	14.07^{**}	2.831^{*}	3.193^{*}
	(16.46)	(1.630)	(5.099)	(1.502)	(1.685)
N	144	144	144	16	16
R^2	0.345	0.736	0.739	0.776	0.771
adj. R^2	0.331	0.730	0.733	0.742	0.735

Table 1: Bubble size indicators across treatments

Notes: * p<0.1, ** p<0.05, *** p<0.01. The indicators in the first three columns are derived from the model. The standard bubble indicators used in the experimental literature are shown in the last two columns. $B_t = Q_t - FV_t$ denotes the absolute bubble size (mean trading price minus fundamental value), B_t/FV_t the relative bubble size, $PRR_t = Q_t/P_t^r$ the Price-to-Rent Ratio. $RAD = \frac{1}{N} \sum_{t=1}^{N} |Q_t - FV_t| / |FV|$ denotes the Relative Absolute Deviation and $RD = \frac{1}{N} \sum_{t=1}^{N} (Q_t - FV_t) / |FV|$, where $FV = \frac{1}{N} \sum_{t=1}^{N} FV_t$, the Relative Deviation; (Stöckl et al., 2010). **Panel A:** The first two rows show the mean values for both treatments, the third row the Z-values from a Mann–Whitney U test. The sample size of each test is N = 16. **Panel B:** Ordinary Least Square regression with standard errors in parentheses (clustered at session level where appropriate, robust otherwise). The dummy variable ξ takes the value one for the treatment ($\xi = 6$), and zero otherwise. Controls are period and gender composition in the first three columns and gender composition in the last two columns.

4.3 Robustness Check 1: Liquidity Constraint

Figure 2 illustrates that the average mean trading price Q_t of the housing asset H_t is similar in both treatments; 28.58 for treatment ($\xi = 2$) and 28.85 for treatment ($\xi = 6$). This section investigates the possibility that this realized trading price results from a binding liquidity constraint. Such an endowment mechanism does not exist in the theoretical model of Huber (2018) where the equilibrium is an interior solution but could, in principle, exist in the experiment. The theoretical maximum absolute bubble size B_t^{max} is limited from below by the fundamental value and from above by the liquidity in the market; see Equation (2.7). Since the endogenously realized price for housing service P_t^r , and thus the fundamental value, are higher in the treatment ($\xi = 6$), the difference in the realized bubble size across treatments could result as a consequence of a binding liquidity constraint. To investigate this possibility, we compute the maximum possible trading price Q_t^{max} for both treatments—the house price that would prevail if all households (after purchasing the consumption good C_t and housing services S_t) would have zero savings in the riskless bond Z_t , and instead, would spend all their remaining budget on housing assets H_t . The calculation is detailed in Appendix C.

Households could have spent maximally $Q_t^{max} = 47.01$ and $Q_t^{max} = 43.26$ experimental currency units per unit of housing asset H_t in treatment ($\xi = 2$) and ($\xi = 6$), respectively. The difference between the upper bound Q_t^{max} and the realized market price Q_t amounts to 18.67 and 14.28 experimental currency units in treatment ($\xi = 2$) and ($\xi = 6$), respectively. The difference between the upper bound Q_t^{max} and the realized trading price Q_t is at least 30% in either treatment. Figure 2 illustrates this difference visually. This gap is too large to be considered a binding liquidity constraint for the realized trading price Q_t . We therefore conclude that the variation in the realized bubble sizes across treatments is attributed to the different exogenous preferences for housing services (and hence the different endogenous fundamental values).

4.4 Robustness Check 2: Cash-to-Asset Ratio (CAR)

Mortgage market conditions and credit availability played an important role for the formation of house price bubbles in the aftermath of the Great Recession (for instance, (Claessens et al., 2009), (Agnello and Schuknecht, 2011), (Igan and Loungani, 2012), and (Schularick and Taylor, 2012)). A number of asset market experiments support this view by providing evidence that the level of cash and consequently the cash-to-asset ratio (CAR) matter for the magnitude of asset prices and bubble formation (for instance, (Smith et al., 1988), (Haruvy and Noussair, 2006), (Caginalp et al., 2001), (Caginalp et al., 1998), (Kirchler et al., 2012), (Noussair and Tucker, 2016), (Weitzel et al., 2019), (Kirchler et al., 2015), (Razen et al., 2017), (Hens and Steude, 2009), and (Kose, 2015)). However, reviewing this literature suggests that the impact of cash holdings depends crucially on the setup of the experimental asset market. A comparison of design differences and similarities between these experimental asset markets and ours is presented in Appendix C. Although our experimental design is somewhat closer to market settings with null results (relatively constant FV and CAR over time, small cash differences across treatments), we nevertheless perform a robustness check. It addresses the concern that slight *endogenously* emerging budget differences drive the treatment effect—instead of differences in the preference for housing services affecting the bubble size through the fundamental value.

Recall that the *initial* cash holdings are identical across treatments. Participants take

their consumption decisions (consumption good C_t and housing services S_t), which determines the FV_t of the housing asset and the remaining budget for the investment decisions. Households also make investment decisions (housing asset H_t and riskless bound Z_t), which determines the housing price Q_t . Hence, if endogenous cash holding differences (that emerge through the consumption decisions C_t and S_t) would drive the treatment effect (the different bubble sizes), then this liquidity effect has to go via the housing price Q_t . The FV_t is already determined by the young households' S_t purchases and is hence not affected. The endogenous differences in cash holdings do not translate into different trading prices in our experiment. Figure 2 shows that the realized trading price Q_t is very similar across treatments and not statistically significantly different (p = 0.574). This result is not surprising, given that the initial cash holdings are identical and the resulting endogenous difference in available cash before the double auction is small on an experimental scale (approximately 9% in our experiment compared to approximately 900% in (Weitzel et al., 2019)).

To further address the concern that the small differences in cash holdings drive the treatment effect, we adjust the trading price in treatment ($\xi = 2$) for the nine percent greater liquidity in the market. We do so by reducing the realized trading price Q_t in the treatment ($\xi = 2$) by 2.7 experimental currency units (= 0.674 · 4). The figure 67.4% corresponds to the share of the cash holdings (= 865.20; after purchasing C_t and S_t) that the average household invests in H_t in treatment ($\xi = 6$).⁴⁰ The number 4 is the theoretical price difference of asset H_t due to cash differences (= 80 remaining budget/ 20 units of H_t).⁴¹ Alternatively, one could adjust the trading price in the treatment ($\xi = 6$), arguing that, if the young households had had the same cash holdings (i.e., 80 experimental currency units more), then they might have spent it on the housing asset. The results are robust to this alternative.

Table 2 presents the same tests for the bubble indicators as in subsection 4.2 but calculated with the adjusted trading price Q^{adj} for the treatment ($\xi = 2$). As can be seen from the non-parametric test results in Panel A and the regression results in Panel B, each bubble measure remains significantly different across treatments. The results confirm that the housing bubble size is significantly smaller in the treatment with a strong preference for

⁴⁰Alternatively, we could take the share of the cash holdings of 940.20 invested in H_t in treatment $\xi = 2$ (60.3%), but we decided on the more conservative robustness test.

⁴¹In equilibrium, the cash difference when entering the double auction would be 80 (= 940 - 860) experimental currency units. The endogenously determined steady state price for one unit of S_t is 2 in treatment ($\xi = 2$), and it is 6 in treatment ($\xi = 6$). Since all parameters and the experimental setup are otherwise identical, the remaining budget when entering the double auction for H_t is larger in treatment ($\xi = 2$) by maximally (6 - 2) $\cdot 20 = 80$ experimental currency units.

housing services. This large difference cannot be explained by an endogenously resulting nine percent larger liquidity in treatment ($\xi = 2$).

	B_t^{adj}	$(B_t/FV_t)^{adj}$	PRR^{adj}	RAD^{adj}	RD^{adj}
Panel A: MWU tests					
Treatment $(\xi = 2)$	19.34	3.071	12.93	3.066	3.066
Treatment $(\xi = 6)$	10.61	0.582	5.070	0.553	0.442
Z	2.100^{**}	3.361^{***}	3.361^{***}	3.361^{***}	3.361^{***}
Panel B: OLS regressions					
Treatment dummy $(\xi = 6)$	-8.347**	-2.503***	-7.899***	-2.513***	-2.624***
	(3.663)	(0.408)	(1.280)	(0.406)	(0.425)
Constant	29.10*	3.003^{*}	12.71^{**}	2.194	2.773
	(16.46)	(1.626)	(5.095)	(1.410)	(1.684)
N	144	144	144	16	16
R^2	0.241	0.672	0.674	0.733	0.732
adj. R^2	0.225	0.665	0.667	0.714	0.713

Table 2: Bubble size indicators controlling for CAR

Notes: * p<0.1, ** p<0.05, *** p<0.01. The indicators in the first three columns are derived from the model. The standard bubble indicators used in the experimental literature are shown in the last two columns. $B_t^{adj} = Q_t^{adj} - FV_t$ denotes the absolute bubble size (mean trading price minus fundamental value), B_t^{adj}/FV_t the relative bubble size, $PRR_t^{adj} = Q_t^{adj}/P_t^r$ the Price-to-Rent Ratio. $RAD^{adj} = \frac{1}{N} \sum_{t=1}^{N} |Q_t^{adj} - FV_t| / |FV|$ denotes the Relative Absolute Deviation and $RD^{adj} = \frac{1}{N} \sum_{t=1}^{N} |Q_t^{adj} - FV_t| / |FV|$ denotes the Relative Absolute Deviation and $RD^{adj} = \frac{1}{N} \sum_{t=1}^{N} |Q_t^{adj} - FV_t| / |FV|$, where $FV = \frac{1}{N} \sum_{t=1}^{N} FV_t$, the Relative Deviation; (Stöckl et al., 2010). The superscript *adj* refers to an adjustment of the trading price in treatment $\xi = 2$ to control for endogenously resulting small differences in market liquidity. **Panel A:** The first two rows show the mean values for each treatments, the third row the Z-values from a Mann–Whitney U test. The sample size of each test is N = 16. **Panel B:** Ordinary Least Square regression with standard errors in parentheses (clustered at session level where appropriate, robust otherwise). The dummy variable ξ takes the value one for the treatment ($\xi = 6$), and zero otherwise. Controls are period and gender composition in the first three columns and gender composition in the last two columns.

5 Conclusion

The dual nature of housing makes it a particular asset. Housing entails a consumption and an investment side, and both are economically relevant from a household perspective. Housing represents a large share of the wealth of private households, and households spend the largest share of their monthly consumption expenses on housing, in all OECD countries. However, the relative importance of housing consumption varies greatly across countries—as does the magnitude of housing bubbles—resulting in a significant negative relationship between housing consumption expenditures and housing bubble sizes. While the existing literature studying determinants for house price volatility and the bubble size focuses exclusively on aspects that work through the investment side of the housing asset, the role of housing consumption has not been explored to the best of our knowledge. This paper aims to fill this gap.

The 'consumption channel' is universally valid as long as the asset under consideration is a consumption good at the same time. For example, artwork, classic cars, and jewelry are simultaneously consumption and investment goods. However, given that housing consumption constitutes the largest consumption expenditure share for most economies, this 'consumption channel' is particularly relevant for housing bubbles.

In our theoretical framework and laboratory experiment, the economy is modeled as an OLG world and the investment and consumption side of housing are explicitly and separately modeled. Furthermore, a market for the traded asset (the house) and a market for the traded asset's dividend (price for housing services) exists, particularly essential for assets that entail a consumption and an investment component. The asset's dividend is thus determined *endogenously* by the consumption demand of households. This framework and a number of novel features (among them, subjects' assignment to markets, the OLG incentive structure) make our experimental design a credible starting point for follow-up studies on experimental (housing) asset markets.

Consistent with the model predictions, our results confirm that housing bubbles are substantially and significantly larger when the demand for housing services (i.e., housing consumption) is weak. The bubble size (both absolute and relative) and the price-to-rent ratio are larger in the treatment with weaker preferences for housing services. Several wellestablished experimental bubble indicators draw the same conclusion. Weak preferences for housing services lead to a significantly lower demand for housing services (relative to other consumption goods), and consequently, a significantly lower dividend and fundamental value for the housing asset; eventually resulting in larger housing bubbles. We can rule out a number of potential alternative channels—including, differences in liquidity or cashto-asset ratios—and are confident in concluding that differences in the dividend and thus fundamental value drive the treatment effect. Furthermore, the bubble size is relatively stable over time, as would be expected in an overlapping generation world.

As mentioned earlier, our design provides an adequate starting point for follow-up studies. For instance, competing policy interventions in the housing market to manage bubbles can be analyzed in our OLG design. Follow-up work may also compare the relative merits of policy interventions that aim to foster housing affordability (e.g., rental subsidies, rental caps, help-to-buy schemes) and their implications for economies' vulnerability to housing bubbles. Furthermore, the OLG structure allows the analysis of how policy interventions affect the young and the old generation (differentially) and how changes in population dynamics (resulting from aging or migration) affect housing bubbles.⁴²

⁴²For ethical reasons, it can be challenging to test the causal effect of policy interventions in the field. A laboratory setup provides a credible alternative. Note that our two-period OLG design can be easily modified to an OLG environment with several periods per lifecycle if needed.

Appendix A: Design features

We ask young households to submit the maximum number of units of $C_{y,t}$ and S_t they want to purchase. Once all young households in the market have submitted their demands, the algorithm checks for availability of the demanded number of units, and the resulting market price for one unit of S_t , P_t^r , as well as the finally purchased units of the consumption good $C_{y,t}$ and housing service S_t are displayed on the screen.

In the case of excess demand, the number of units demanded by each young household is reduced proportionately until it matches the supply of units of $C_{y,t}$ and S_t . Excess demand existed for $C_{y,t}$ and S_t in 234 and 284 out of a total of 288 market encounters, respectively. Note that only relative demand for $C_{y,t}$ and S_t matter for price formation, and there is no reason to believe that the proportional assignment (in the case of excess demand) matters differently for $C_{y,t}$ versus S_t .

Parameter	Calibration	Explanation
Supply		
$\frac{C_t^{supply}}{S_t^{supply}}$	= 20 units of consumption good = 20 units of housing services	In each market (A & B) and period In each market (A & B) and period
H_t^{supply}	= 20 units of housing asset	In each market (A & B) and period
Endowment		
$\frac{E_t}{E_{o,1}}$	250 experimental currency units 50 experimental currency units in Z_1 5 units of H_1	Young household at beginning of lifecycle Old household in period 1 Old household in period 1
Prices		
$\begin{array}{c} P_t \\ P_t^r \\ Q_t \\ R \\ \text{Dividend} \end{array}$	$ \begin{split} &= 1 \\ &= \xi * \frac{C^{demanded}}{S^{demanded}} \\ &\text{determined in a double auction} \\ &= 1.05 \\ &= P_t^r \text{ if } S_t^{demand} = S_t^{supply} \\ &= P_t^r \text{ if } S_t^{demand} > S_t^{supply} \\ &= P_t^r * \frac{S_t^{demand}}{S_t^{supply}} \text{ if } S_t^{demand} < S_t^{supply} \end{split} $	Price for one unit of C_t (numeraire) Relative price for one unit of S_t Average price for one unit of H_t Interest rate on riskless bond holding Z_t Rental income generated by asset H_t
Utility		
$U(\xi = 2)$ $U(\xi = 6)$	$= log(C_{y,t}) + 2log(S_t) + log(C_{o,t+1}) = log(C_{y,t}) + 6log(S_t) + log(C_{o,t+1})$	Utility in treatment $\xi = 2$ Utility in treatment $\xi = 6$
Demanded units		
$\begin{array}{c} X_{i,t} \\ X_{i,t} \end{array}$	$ \begin{aligned} &= X_{i,t} \text{ if } \sum_{i=1}^{4} X_i^{demand} \leq X^{supply} \\ &= X_{i,t}^{adj} \text{ if } \sum_{i=1}^{4} X_{i,t}^{demand} > X_t^{supply} \end{aligned} $	No aggregate excess demand Aggregate excess demand

Table A1: 1	Parameter	choices
Table A1: 1	Parameter	choices

Notes: $X_{i,t}$ denotes the *actual* individual demand of subject *i* in period *t* of variable $X \in \{C_t, S_t, H_t\}$, while X_t^{supply} denotes aggregate supply of X_t . $X_{i,t}^{adj}$ denotes the individual *adjusted* demand of subject *i* in period *t* in case of aggregate excess demand and is given by $X_{i,t}^{adj} = X_{i,t} \frac{X_t^{supply}}{\sum_{i=1}^{4} X_{i,t}^{admand}}$, hence individual demand gets proportionally adjusted.

Young households purchase consumption goods $C_{y,t}$ and housing services S_t in a graph on the decision screen and can simulate other young households' purchase decisions. Alternative procedures to determine the market demand and market price for housing services S_t include the elicitation of full demand schedules or call markets. Ideally, we would have employed demand schedules where households are price-takers. Note however that—in the model and in our design—young households demand two consumption goods simultaneously, $C_{y,t}$ and S_t , which both affect the price of housing services P_t^{r} .⁴³ This complicates the implementation of full demand schedules and call markets substantially. Due to time constraints (the experiment lasted 2.5 hours on average) and the importance of the consumption utility, we decided for the described procedure, which we believe was a good feasible solution. Our setup allowed subjects to form adequate point estimates about P_t^r : Subjects first performed four practice periods, and during the incentivized periods, more than 75% of subjects revealed a stable relative demand for $C_{y,t}/S_t$.⁴⁴

Period		1	2	3	4	5	6	7	
		young	old	young	old	young	old	young	
Cohort I	Lifecycle	-	1	2	2	;	3		
(o subjects)	(8 subjects) Market		A or B (random)		A or B (random)		A or B (random)		Etc.
		old	young	old	young	old	young	old	
Cohort II	Lifecycle			1	2	2	;	3	
(8 subjects)	Market	A or B	A c (rand	or B dom)	A c (rand	or B dom)	A c (rand	or B dom)	

Figure A1: Chronological order of the experiment

 $^{{}^{43}}P_t^r = p_t^r$, since the price for C_t , $P_t = 1$, is the numeraire.

⁴⁴We eye-balled the individual time trends of each subject. Figures are available upon request.

Appendix B: Additional Results

Table B1 provides the average bubble size (for all five bubble size indicators) at a session level. Figure B1 plots the mean trading price and the fundamental value for each session separately. Figure B2 provides an even more detailed view and illustrates mean trading price and the fundamental value for each of the two markets within one session.

Constant Bubble Size

In the OLG model, households live for two periods and only buy and sell the housing asset H_t once in their lifetime. We compare two deterministic steady states of the model, one with the key model parameter $\xi = 2$ and one with $\xi = 6$. In each deterministic steady state, the absolute and relative bubble size and the price-to-rent ratio are constant and larger in the steady state with a weaker preference for housing services ($\xi = 2$). In the experiment, subjects play several lifecycles but are incentivized to treat each lifecycle as if it was the only one they live: only one lifecycle is chosen randomly and paid out. If subjects are correctly incentivized (as if they live for one lifecycle only), one would not expect any boom-bust cycles in the trading price or the bubble size. Subjects would decide what the optimal decision is and replay this decision in every lifecycle.

As predicted by the OLG model, the absolute and relative bubble size as well as the trading price Q_t do not display any boom-bust cycles over time. This finding relates to Hoshihata et al. (2017), who refer to such bubbles of constant magnitude as "flat bubbles". Empirical work often measures bubbles in asset prices by boom-bust cycles. Many experimental asset markets display boom-bust cycles as well. The large majority of these experiments are based on finite horizons—in contrast to the indefinite two-period overlapping generation design that we use in our experiment. In the experimental asset price literature, flat bubbles are a rare event. Aside from Hoshihata et al. (2017), Kopanyi-Peuker and Weber (2021) also observe flat bubbles that continue until the market terminates. The underlying asset market designs of both aforementioned studies differ from our OLG structure. However, both market designs share a key characteristic with our design: a long time horizon. In contrast to typical asset market experiments, the time horizon in Kopanyi-Peuker and Weber (2021) and our study is indefinite, while the time horizon is very long but fixed to 100 periods in Hoshihata et al. (2017).

Session	B_t	B_t/FV_t	PRR_t	RAD	RD
S1 Average	15.51	2.49	11.09	2.49	2.49
S4 Average	16.21	2.55	11.35	2.55	2.55
S5 Average	27.96	4.42	17.28	4.41	4.41
S7 Average	28.66	4.48	17.53	4.48	4.48
S9 Average	22.90	3.49	14.30	3.49	3.49
S12 Average	13.24	2.11	9.81	2.10	2.10
S13 Average	21.53	3.36	13.97	3.36	3.36
S16 Average	31.00	5.09	18.92	5.07	5.07
Treatment $\xi = 2$ Average	22.03	3.50	14.28	3.49	3.49
S2 Average	11.55	0.60	5.15	0.60	0.60
S3 Average	12.78	0.68	5.32	0.68	0.68
S6 Average	11.32	0.64	5.38	0.64	0.64
S8 Average	2.11	0.11	3.58	0.14	0.11
S10 Average	28.21	1.59	8.13	1.58	1.58
S11 Average	-4.68	-0.25	2.42	0.27	-0.26
S14 Average	10.28	0.58	5.11	0.59	0.59
S15 Average	13.30	0.71	5.45	0.71	0.71
Treatment $\xi = 6$ Average	10.61	0.58	5.07	0.65	0.58

 Table B1: Bubble Size Indicators on Session Level

Notes: The indicators in the first three columns are derived from the model. The standard bubble indicators used in the experimental literature are shown in the last two columns. $B_t = Q_t - FV_t$ denotes the absolute bubble size (mean trading price minus fundamental value), B_t/FV_t the relative bubble size, $PRR_t = Q_t/P_t^r$ the Price-to-Rent Ratio. $RAD = \frac{1}{N} \sum_{t=1}^N |Q_t - FV_t| / |FV|$ denotes the Relative Absolute Deviation and $RD = \frac{1}{N} \sum_{t=1}^N (Q_t - FV_t) / |FV|$, where $FV = \frac{1}{N} \sum_{t=1}^N FV_t$, the Relative Deviation; (Stöckl et al., 2010).



----- (1 + 60%) * FV ------ Generating and the second seco

Figure B1: Mean Trading Price Q and FV (by Session Average). Notes: ξ denotes the treatment parameter; the preference for housing services S_t (relative to other consumption goods C_t).



Figure B2: Mean Trading Price Q and FV (by Market and Session). Notes: ξ denotes the treatment parameter; the preference for housing services S_t (relative to other consumption goods C_t).

Appendix C: Robustness Checks

Robustness Check 1: Liquidity Constraint

The maximum possible trading price Q_t^{max} in each treatment is calculated as follows. To compute the available budget when entering the double auction for housing assets H_t , we first need to subtract aggregate consumption spending on C_t (= $P_t \cdot 20$) and S_t (= $P_t^r \cdot 20$) from the market liquidity E_t (= 4.250). Households have an average remaining budget of 940.20 and 865.20 experimental currency units in treatment ($\xi = 2$) and ($\xi = 6$), respectively.⁴⁵ This budget difference is small on an experimental scale. We obtain the maximum price Q_t^{max} that households could spend on the housing asset H_t by dividing the remaining budget in the economy by the available housing assets H_t (= 20 units).

Robustness Check 2: Cash-to-Asset Ratio (CAR)

A finite-horizon with a decreasing FV over time characterizes the experimental asset market of the seminal paper by Smith et al. (1988) (denoted by SSW). The dividend payments each period increase the cash in the economy, which—coupled with the decreasing FV increases the CAR over time by construction. SSW find that experimental asset prices deviate strongly from fundamental values.⁴⁶ In finite-horizon setups with constant FV, Noussair and Tucker (2016) and Weitzel et al. (2019) find that high initial cash holdings and a high constant CAR are associated with higher prices and bubbles.⁴⁷

In contrast to the aforementioned studies, our experimental design consists of an indefinite horizon and an endogenous, relatively constant FV of the housing asset. For indefinite-horizon asset markets with constant FV, the evidence on the impact of cash holdings on asset prices is rather scarce and mixed. The setup of Hens and Steude (2009) is characterized by an increasing CAR over time; interestingly three out of four markets do not display any over-evaluation of the asset. More generally, Kose (2015) finds that overpricing depends on the existence of a redemption value in such a context. They document higher prices and larger overpricing when the constant FV does not include a redemption value. Our experimental design is similar in that the horizon is indefinite, the (endoge-

⁴⁵The average price for housing services S_t is 1.99 and 5.74 experimental currency units in treatment ($\xi = 2$) and ($\xi = 6$), respectively.

⁴⁶Haruvy and Noussair (2006), Caginalp et al. (2001), and Caginalp et al. (1998) show that high initial cash-to-asset ratios increase asset prices strongly in such a SSW-settings and lead to larger asset bubbles. Kirchler et al. (2012) find that bubbles emerge when a decreasing fundamental value is coupled with an increasing CAR (as in the SSW setup). However, when the FV follows a constant time trajectory, they find no effect of cash holding levels or the CAR on asset prices.

⁴⁷Also the inflow of new traders with new cash and a consequently increasing CAR can trigger large asset price increases (Kirchler et al., 2015), but not always (Razen et al., 2017).

nous) FV is constant (without redemption value), and cash holdings (as well as the CAR) are constant. Differences include the endogenous nature of the FV and the overlapping–generation design. We contribute to this literature by showing that an indefinite–horizon asset market with an OLG structure and endogenous constant FV (without redemption value) features consistently overpricing.

In our experiment, the *initial* cash is the same for both treatments (and constant over time). However, we observe that the young's remaining budget after purchasing C_t and S_t and before entering the double auction for H_t differs slightly across treatments. We investigate the impact of the endogenously emerging difference in cash holdings on the trading price Q_t . As explained in the previous subsection, households have an average remaining budget of 940.20 and 865.20 experimental currency units in treatment ($\xi = 2$) and ($\xi = 6$), respectively. In our setup, the cash difference on the asset market across treatments is rather small (approximately 9%) compared to Weitzel et al. (2019) where liquidity across treatments differs by approximately 900%.

Robustness Check 3: Assumptions about Expectations

For robustness purposes, we measure the *expected* fundamental value as in Equation (4.1) using two alternative adaptive expectation formations. Our results are robust to these alternative types of expectation formation and are available upon request.

We call the first alternative *naive* households. We assume that naive households expect the dividend to be constant and equal to the first realization. The naive households calculate and expect the following fundamental value:

$$FV_t^{naive} \equiv E_t \left\{ \sum_{j=1}^{\infty} \left(\frac{1-x}{R} \right)^j p_{t+j}^r \right\}$$

= $\frac{1}{R} (1-x) p_{t=1}^r + \frac{1}{R^2} (1-x)^2 p_{t=1}^r + \frac{1}{R^3} (1-x)^3 p_{t=1}^r + \dots$
= $p_{t=1}^r \left(\frac{1-x}{R-(1-x)} \right) \quad \forall \quad t.$ (C1)

The second alternative are *myopic* households. We assume that myopic households observe the dividend payment in each period and expect all future dividends to be equal to the current realization. In all periods, myopic households update their belief and expect that all future dividends will be equal to the currently realized dividend. The myopic households calculate and expect the following fundamental value:

$$FV_t^{myopic} \equiv E_t \left\{ \sum_{j=1}^{\infty} \left(\frac{1-x}{R} \right)^j p_t^r \right\}.$$
 (C2)

Robustness Check 4: Median (instead of Mean) Trading Price

Figures C1–C4 replicate our key findings when we use the median trading price of the housing asset instead of the mean trading price. The treatment effect is statistically significant for all five bubble indicators.



Figure C1: Average Median Trading Q_t^m Price and FV (by Treatment). Notes: ξ denotes the preference for housing services S_t (relative to consumption C_t).



Figure C2: Relative Bubble Size B_t^m/FV_t (by Treatment and Session). Notes: ξ denotes the preference for housing services S_t (relative to consumption C_t); $B_t^m = Q_t^m - FV_t$, where Q_t^m denotes the median trading price.



Figure C3: Median Trading Price Q_t^m and FV (by Treatment and Session). Notes: ξ denotes the preference for housing services S_t (relative to consumption C_t).



Figure C4: Median Trading Price Q_t^m and FV (by Market and Session). Notes: ξ denotes the preference for housing services S_t (relative to consumption C_t).

Robustness Check 5: Experimental Bubble Indicators

We employ a number of alternative experimental bubble indicators, namely the following:

Price Amplitude (PA)

$$PA_{King} = \frac{max_t(Q_t - FV_t) - min_t(Q_t - FV_t)}{FV_1}$$
(C3)

is defined as the difference between the peak and the trough of the period trading price relative to the fundamental value, normalized by the initial fundamental value in period one. A high PA suggests large price swings relative to the fundamental value. This measure was first proposed by King et al. (1991).

Total Dispersion (TD)

$$TD = \sum_{t} |Q_t - FV_t| \tag{C4}$$

is the sum of all period absolute deviations of median prices from the fundamental value and thus a measure of the magnitude of overall mispricing. This measure was first introduced by Haruvy and Noussair (2006). The TD measures the difference between the trading price and the fundamental value, and is hence similar to the PA measure. However, the TD measure is more complete in the sense that it does not only measure the difference between the maximum and minimum deviation from the fundamental value.

Average Bias (AB)

$$AB = \frac{\sum_{t} (Q_t - FV_t)}{T} \tag{C5}$$

was first introduced by Haruvy and Noussair (2006) and is calculated by the sum of all period absolute deviations of median trading prices from fundamental value, normalized by the total number of periods T. Hence, it is an indicator of the average per-period deviation of trading prices from the fundamental value.

The first column of Table (C1) shows the Price Amplitude (PA). According to this measure, the bubble in the treatment "Weak preference for housing services ($\xi = 2$)" is on average almost three times as large. The second column of Table (C1) shows the Total Dispersion (TD) measured by the sum of all period absolute deviations of median trading prices from the FV. According to this measure, the bubble is significantly larger in the treatment "Weak preference for housing services ($\xi = 2$)". The third column of Table (C1) shows the indicator Average Bias (AB); AB averages the sum of all median price deviations from the FV. These three additional experimental bubble indicators each reveal that the bubble size is substantially larger for the treatment "Weak preference for housing services ($\xi = 2$)". Table C2 and C3 show that this difference between the two treatments is statistically significant.

Session	PA	TD	AB
S1 Average	0.48	140.34	15.59
S4 Average	1.82	159.42	17.71
S5 Average	2.41	252.50	28.06
S7 Average	0.59	258.90	28.77
S9 Average	0.42	200.99	22.33
S12 Average	4.21	114.72	12.75
S13 Average	0.64	194.90	21.66
S16 Average	0.96	281.23	31.25
Treatment $\xi = 2$ Average	1.44	200.38	22.26
S2 Average	0.31	99.99	11.11
S3 Average	0.00	110.00	10.05
0011010260	0.38	113.89	12.65
S6 Average	$\begin{array}{c} 0.38\\ 0.37\end{array}$	$113.89 \\ 103.05$	$12.65 \\ 11.45$
S6 Average S8 Average	$0.38 \\ 0.37 \\ 0.27$	$ \begin{array}{r} 113.89 \\ 103.05 \\ 18.16 \end{array} $	$12.65 \\ 11.45 \\ 2.02$
S6 Average S8 Average S10 Average	$\begin{array}{c} 0.38 \\ 0.37 \\ 0.27 \\ 1.29 \end{array}$	$ \begin{array}{r} 113.89 \\ 103.05 \\ 18.16 \\ 252.76 \\ \end{array} $	$ 12.65 \\ 11.45 \\ 2.02 \\ 28.08 $
S6 Average S8 Average S10 Average S11 Average	$\begin{array}{c} 0.38 \\ 0.37 \\ 0.27 \\ 1.29 \\ 0.47 \end{array}$	$113.89 \\103.05 \\18.16 \\252.76 \\-48.62$	12.65 11.45 2.02 28.08 -5.40
S6 Average S8 Average S10 Average S11 Average S14 Average	$\begin{array}{c} 0.38 \\ 0.37 \\ 0.27 \\ 1.29 \\ 0.47 \\ 0.36 \end{array}$	113.89 103.05 18.16 252.76 -48.62 92.29	$12.65 \\ 11.45 \\ 2.02 \\ 28.08 \\ -5.40 \\ 10.25$
S6 Average S8 Average S10 Average S11 Average S14 Average S15 Average	$\begin{array}{c} 0.38 \\ 0.37 \\ 0.27 \\ 1.29 \\ 0.47 \\ 0.36 \\ 0.46 \end{array}$	113.89 103.05 18.16 252.76 -48.62 92.29 122.79	$12.65 \\ 11.45 \\ 2.02 \\ 28.08 \\ -5.40 \\ 10.25 \\ 13.64$

Table C1: Additional Experimental Bubble Indicators

Table C2: Difference of Bubble Size Indicators between Treatments

	PA	TD	AB
Δ mean	0.95***	106.09^{***}	11.79***
Z	3.361	2.626	2.626
N	16	16	16

Notes: The values represent the difference in means of the two treatments and Z-values from a Mann–Whitney U test. * p < 0.10, ** p < 0.05, *** p < 0.01. PA (Price Amplitude)= $max(Q_t - FV_t)/FV_1 - min(Q_t - FV_t)/FV_1$; (Porter and Smith, 1995). TD (Total Dispersion)= $\sum_{t=1}^{N} |Q_t^m - FV_t|$ and AB (Average Bias)= $\frac{1}{N} \sum_{t=1}^{N} (Q_t^m - FV_t)$; (Haruvy and Noussair, 2006). FV_t denotes the fundamental value, Q_t the mean trading price and Q_t^m denotes the median trading price.

PA (Price Amplitude) = $max(Q_t - FV_t)/FV_1 - min(Q_t - FV_t)/FV_1$; (Porter and Smith, 1995). TD (Total Dispersion) = $\sum_{t=1}^{N} |Q_t^m - FV_t|$; (Haruvy and Noussair, 2006). AB (Average Bias) = $\frac{1}{N} \sum_{t=1}^{N} (Q_t^m - FV_t)$; (Haruvy and Noussair, 2006). FV_t denotes the fundamental value, Q_t the mean trading price and Q_t^m denotes the median trading price.

	PA	TD	AB
	(1)	(2)	(3)
ξ	-1.021*	-101.8**	-11.31*
	(0.524)	(35.50)	(3.945)
Constant	-0.086	295.4^{*}	32.82*
	(1.511)	(154.48)	(17.16)
\overline{N}	16	16	16
R^2	0.253	0.385	0.385
adj. R^2	0.138	0.290	0.290

Table C3: Impact of ξ on Bubble Size Indicators

Notes: Ordinary Least Square regressions (robust std. errors in parentheses). * p < 0.10, *** p < 0.05, *** p < 0.01. The dummy ξ is equal to one for the treatment "Strong preference for housing services ($\xi = 6$)", and zero otherwise. We control for the session's gender composition. PA (Price Amplitude)= $max(Q_t - FV_t)/FV_1 - min(Q_t - FV_t)/FV_1$; (Porter and Smith, 1995). TD (Total Dispersion)= $\sum_{t=1}^{t} |Q_t^m - FV_t|$ and AB (Average Bias)= $\frac{1}{N} \sum_{t=1}^{N} (Q_t^m - FV_t)$; (Haruvy and Noussair, 2006). FV_t denotes the fundamental value, Q_t the mean trading price and Q_t^m denotes the median trading price.

Robustness Check 6: Reaction to the Treatment Parameter ξ

In the main text of the paper, we focus on the impact of the preference parameter ξ on the house price bubble size. In this section, we investigate the reaction to the treatment parameter ξ on the consumption decisions C and S in further detail. We find that the purchase decisions for the consumption good C, and thus the ratio of housing services over consumption S/C, differ significantly across treatments. As the model predicts, households purchase significantly fewer units of C in treatment ($\xi = 6$) compared to the treatment ($\xi = 2$) (z = 3.123 and p = 0.0018; two-sided Mann–Whitney U test). The difference in the ratio of housing services over consumption S/C is significantly larger with $\xi = 6$ compared to the treatment ($\xi = 2$) (z = -3.123, p = 0.0018, two-sided Mann–Whitney U test).

Note that, to achieve market clearing, we require the total demand of C and S to be equal to or smaller than the exogenous supply of 20 units, respectively. We observe excess demand in most market encounters. When looking at the actually requested units of Cand S, before adjusting for excess demand, we find that the difference in the ratio S/C is highly significant across treatments (z = -2.71, p = 0.0063; two-sided Mann–Whitney U test).⁴⁸

Recall that the price of C is the numeraire and is set equal to one. The relative price of housing services S is determined endogenously by the relative consumption choices Cand S of the households, and determines in turn the dividend of housing asset H (the steady state model prediction is $P^r = 2$ in the case of $\xi = 2$, and $P^r = 6$ for the treatment $\xi = 6$). Our experimental data is very close to the model's predictions; the realized price for housing services is $P_{\xi=2}^r = 1.99$ and $P_{\xi=6}^r = 5.74$, respectively. The relative price for housing services P^r is significantly different across treatments (z = 3.363 and p = 0.0008; two-sided Mann–Whitney U test). Therefore, we conclude that the participants do react to the treatment parameter ξ .

⁴⁸For the raw data, the difference across treatments becomes larger and remains in line with the model's prediction. The treatment differences are slightly significant for C (z = 1.785, p = 0.0742 for C and z = -1.5, p = 0.1152 for S; two-sided Mann–Whitney U test).

Appendix D: Instructions & Screenshots

Written Instructions

Welcome and General Instructions

Thank you for participating in this experiment. You are taking part in an experiment involving decisions on experimental groups.

Please read these instructions carefully; they will help you make appropriate decisions. You will receive 5 euros for participating in this experiment and another 5 euros for finishing the experiment. Furthermore, you will earn money depending on your decisions and the decisions of other participants during the experiment. Depending on your own and other participants' decisions you may earn a considerable amount of money.

At the end of the experiment, your earnings will be immediately paid out in cash.

Questions

Please feel free to raise your hand and ask the experimenter(s) any question you may have at any time during the experiment.

Please do not talk to other participants until the experiment is over. During the experiment, the use of cell phones is prohibited.

Overview over the Experiment

In this experiment, you will play several "lifecycles". A lifecycle consists of two periods: In the first period of a lifecycle, you are "Young". In the second period of a lifecycle, you are "Old".

In each lifecycle, you can earn Happiness Points, which will depend on your consumption and investment decisions when "Young" and "Old" as well as on other participants' decisions.

You will play several **independent** lifecycles. In each lifecycle, the decisions when "Young"

and "Old" will be the same.

Objective in each lifecycle

Your objective in each lifecycle is to earn as many Happiness Points as possible with your available budget. You earn Happiness Points by purchasing consumption good C and housing service S. Your final budget at the end of the lifecycle will also be transformed into Happiness Points. The number of Happiness Points will be transformed into euros at the exchange rate of 1 Happiness Point = 3 euros.

When "Young," you can use your budget to purchase consumption good C and housing service S. You can also invest in the housing asset H. In case you do not spend all your money on S, C and H, your remaining budget will remain in your bank account B and receive automatically an interest rate payment. Your purchase of consumption good C and housing service S gives you immediately Happiness Points.

When "Old," your housing asset H (you bought when "young") provides a dividend (= return) and a potential profit from reselling it at a higher price to the next young generation. The remaining budget in your bank account B provides a fixed interest. After the period of being "Old," your total returns from housing asset H and bank account B will be transformed into Happiness Points.

Decisions in a lifecycle

Remember: A **lifecycle** consists of two periods: In the **first period** of a lifecycle, you are "**Young**". In the **second period** of a lifecycle, you are "**Old**". **Each period** is split into **two stages**, respectively.

When you are <u>"Young</u>": In stage (1), you decide how many units of consumption good C and how many units of housing service S you want to purchase. In stage (2), you can ask for units of the housing assets H with the remaining budget in a double auction.

When you are <u>"Old"</u>: In stage (1), you can sell your purchased units of the housing assets H (if you have purchased any) in a double auction to the new "Young". In stage (2), you will be informed of your total returns from housing assets H and your bank account B and

you will receive a summary of your lifecycle decisions and the corresponding Happiness Points.

Decisions when being "Young"

You will receive a budget of 250 EURUX, which will be deposited in your bank account. You can use this money to buy consumption good C, housing services S, and housing assets H.

Stage (1) when young

At the top of the screen you'll see a graph with the different combinations of consumption good C (x-axis) and housing services S (y-axis) that you can buy. The graph shows different colors for each combination of consumption C and housing services S chosen. The color map goes from red to yellow to green. The greener the color the more Happiness Points you receive for the specific combination of C and S. The more red the area, the fewer Happiness Points you receive for the corresponding combination of C and S. The formula behind this is: Happiness Points (from C and S) = log(C) + 2*log(S).

You can move the red point in the upper graph to the left. The red point represents your choice of C and S. On the right, you see how many Happiness Points you would receive for this particular combination of C and S. You can try any combination of C and S units and as many combinations as you wish.

The price for one unit of C is fixed and equal to 1 EURUX. Each of the different **combinations of C and S** defines a **price for housing services S**. The price will depend on the combination-choice of all "Young" participants in your group.

The graph with the blue point on the left of the screen helps you to understand what the relative price of housing services might be. The blue point represents the (simulated!) average choice by the other "Young" participants in your group. Note that this is just a simulation and not the final choice of the other "Young" in your group. The simulated price will be displayed on the right side of the screen. Notice that this information is only a potential (simulated) price. The actual price will be computed based on all group members' choices. You will receive information on the total number of Happiness Points and your remaining bank account balance for the chosen combination of C and S units.

Once you have decided on a combination of C and S units on the graph, you submit your final decision by clicking on the button "Submit".

Note that, for all Young in your group, the total available amount of C and S is 20 units, respectively. You input the very maximum amount you would like to purchase. You may end up purchasing less than your desired amount. If the total demand for consumption good C and housing services S in the group is in excess of what is available, you may find yourself able to purchase only a fraction of the units you requested. Each Young's purchased units of C and S will be reduced proportionally to the requested amounts.

After all participants have submitted their consumption and housing services decisions, the price for S will be computed. Your spending on C and S will be debited from your bank account.

The computer will check that every Young is able to pay the purchased units of C and S at the calculated price of C and S. Once everybody is set, you continue to Stage (2).

Stage (2) when young

In Stage (2) when "Young," you have to decide how many assets H you want to buy. The dividend will depend on the decisions with regard to the purchase of C and S made by the future "Young"—that is, the "Young" when you will be "Old". Before buying the housing assets H when "Young," you will find a screen where you can choose different combinations of C and S to simulate the choice made by the future "Young" and its implications for the dividend. The graph will help you get an idea about the dividend.

You can buy as many housing assets H as you wish and as your available budget allows. Your available bank account balance (after having purchased C and S) will be displayed in the upper part of the screen. Below that information, you will see the current number of housing assets H that you hold. Both are instantly updated each time you buy an asset. You will have 3 minutes to buy the assets H. When you are "Young," you will only be able to **buy** assets H. You will not be able to sell them. You can buy assets H from the current "Old" in your group. You will be able to do so in two ways.

<u>First</u>, you can **initiate a purchase** of an asset by **submitting an offer to buy (a price for which you want to buy a unit of asset H)**. If you have money (EURUX) in your bank account and would like to buy an asset, you can initiate the purchase by submitting an offer to buy. Note that the offer cannot be larger than your available budget.

After entering a number in the text area "Enter offer to buy" click on the red button labeled "Submit offer to buy". Immediately, in the column labeled "Offers to buy", you will see a list of numbers ranked from low to high. These numbers are the prices at which all "Young" in your group are willing to buy an asset in this period. The offers to buy will be executed once they are accepted by one of the current "Old" in your group.

On the trading screen, your own offers are marked in blue; others' offers are in black. If you want to buy more assets H, repeat this process.

<u>Second</u>, you can realize a purchase of assets by accepting an offer to sell (accepting a price for one unit of H) submitted by a participant who is currently "Old".

If you have enough money in your bank account, you can buy an asset at one of the prices listed in the "Offers to sell" column, which contains all the offers submitted by participants in the Old role. You buy an asset by selecting one of the others' offers and then clicking on the red button "Buy". The best offer is highlighted in deep blue.

Whenever an offer is accepted, a transaction is executed. Immediately when you accept an offer to sell, you realize a purchase and the number of EURUX in your bank account goes down by the trading price. At the same time, your trading partner realizes a sale and the balance of his/her bank account increases by the trading price. Similarly, your number of assets H goes up by one unit and your trading partner's number of assets H goes down by one unit.

In each group, there will be 20 units of housing assets H (owned by the "Old" in your

group). Assets not sold in the double auction are distributed equally among all "Young" in your group (or until the budget of all "Young" is zero) at a punishment price equal to 1.5 times the median price. To calculate the median price in your group you order all sale prices from lowest to highest and pick the price that is in the middle.

Your remaining bank account balance—that is, the budget that you did not spend on consumption good C, housing service S, and housing asset H—will stay in your bank account B and you will receive an interest rate payment of 5%.

Decision when being "Old"

At the beginning of the "Old" phase of the lifecycle, you receive the interest rate payment in your bank account B; it will be deposited in your bank account.

You will receive a dividend for the housing assets H that you bought when "Young" (if any) and the selling price for your housing assets H. How the dividend and the selling price for H are determined is explained below.

Stage (1) when old

When you are "Old," you will only be able to **sell** the assets H that you purchased when you were "Young" in the same lifecycle. You can sell assets H to the current "Young" in your group. Note that you can only sell as many assets H as you hold. You will have 3 minutes to sell all your assets H. Note that you should sell all your assets H, otherwise you will be punished. You will be able to sell assets H in two ways:

<u>First</u>, you can initiate a sale of assets by submitting an offer to sell (you propose a price for which you want to sell one unit of asset H).

You can enter a number (integer) in the text area labeled "Enter offer to sell" in the first column and then click on the button "Submit offer to sell". A set of numbers will appear in the column labeled "Offers to sell". Each number corresponds to an offer from one of the participants who is currently "Old" in your group. Your own offers are shown in blue; others' offers are shown in black. The offers to sell are ranked from high to low. Each offer introduced corresponds to **one single asset**. Note that by submitting an offer to sell, you

initiate a sale, but the sale will not be executed until someone accepts it.

If you want to sell more of your assets H, repeat this process.

Second, you can realize a sale of an asset H by accepting an offer to buy (accepting a price a "Young" is willing to buy an asset H for).

The highest (best) price currently listed in the column of "Offers to buy" is highlighted in deep blue.

Again, a transaction is executed whenever an offer to buy is accepted. If you accept an offer to buy posted by an other participant, you realize a sale and as a result the amount of EURUX in your bank account increases by the trading price. At the same time, your trading partner realizes a purchase and the balance of his/her bank account decreases by the trading price. Similarly, your number of assets H goes down by one unit and your trading partner's number of assets H goes up by one unit.

For all housing assets H that you do not sell you will be punished. You lose your unsold assets H and you will only receive 50% of the median price that was realized during this period in your group. To calculate the median price you order all sale prices from lowest to highest and pick the price in the middle.

Stage (2) when old

Your total budget when being "Old" includes the remaining bank account balance B plus interest payments, as well as the dividend for your housing assets H and the price at which you sell the housing assets H that you had purchased when being "Young".

Summary of the Lifecycle

You will see a summary of your decisions in the corresponding lifecycle on the screen:

- How many units of C and S you bought in that lifecycle and the respective prices;
- How many units of H you bought;
- The median price of housing asset H of all sold H;

- The dividend of asset H you received when "Old";
- The price for which you have sold the purchased assets H;
- The return on your bank account B;
- The number of Happiness Points you received for this lifecycle.

From this information, your final budget when "Old" will be calculated (in EURUX) and transformed into Happiness Points at the following exchange rate: Happiness Points (from H and B) = log (EURUX).

History screen

To help you with the decisions, you will find on the decision screens when "Young" a button labeled "History". If you click this button, you move to a history screen but can return at any time to the decision screen. On this new screen you will find a summary of your decisions concerning C, S, H, and B as well as the corresponding Happiness Points you received in all previous periods of this experiment. Furthermore, you will find a summary of the median price per housing asset H and the average dividend per housing asset H in all previous periods of this experiment.

Assignment to group A or B

In total, 16 participants participate in this experiment including yourself. All 16 participants will be assigned randomly to Cohort I or Cohort II at the beginning of the experiment—that is, before period 1 starts. You will be informed whether you belong to Cohort I or Cohort II. All participants will remain in the assigned cohort for the entire experiment. A total of 8 participants will form Cohort I and 8 participants will form Cohort II.

At the beginning of period 1, 4 members of each cohort will be randomly assigned to Group A and the other 4 members of each cohort will be assigned to Group B.

In period 1, Cohort I will be "Young" and Cohort II will be "Old", and will make decisions accordingly. To start, each member of Cohort II will be endowed with 5 units of housing assets H and 50 EURUX in their bank account.

In period 2, Cohort I will be "Old" and Cohort II will be "Young". Cohort I will remain in the *same* group (A or B) as in period 1. A total of 4 members of Cohort II will be randomly assigned to Group A and the other 4 members will be assigned to Group B.

In period 3, Cohort I will be "Young" and Cohort II will be "Old". Cohort II will remain in the *same* group (A or B) as in period 2. A total of 4 members of Cohort I will be randomly assigned to Group A and the other 4 members will be assigned to Group B.

Etc.

Chronological order of the experiment

Remember that one lifecycle will be chosen randomly and you will be paid according to your Happiness Points in **that** lifecycle. Cohort I's lifecycle 1 consists of periods 1 and 2, its lifecycle 2 consists of periods 3 and 4, etc. Cohort II's lifecycle 1 consists of periods 2 and 3, its lifecycle 2 consists of periods 4 and 5, etc.

If a cohort is "Old" in the last period of the experiment, that lifecycle is complete and enters the lottery of the randomly selected lifecycle for payment. If a cohort is "Young" in the last period of the experiment, that lifecycle is not complete and does not enter the lottery of the randomly selected lifecycle for payment.

In the graphs and tables attached, you find a summary of the experiment.

There will be two sequences of the just described experiment: One trial sequence with four periods, which does not enter the lottery for payment. It is there to help you get familiar with the experiment. Then there will be a sequence out of which one lifecycle will be chosen randomly at the end of the experiment and paid out.

The experiment ends after each period with a probability of 20%. We have thrown a tensided dice to determine the number of periods, whereby the numbers 0 and 1 indicated ending the experiment and the numbers 2 through 9 indicated continuing the experiment.



Figure C5: Decision screen when "Young": Stage 1 (consumption C and housing services S).



Figure C6: Screen when "Young": Stage 2 (simulation of dividend).



Figure C7: Decision screen when "Young": Stage 2 (housing asset H).



Figure C8: Decision screen when "Old": Stage 1 (housing asset H)

Handouts: Graphs and Table

Decisions in a lifecycle



Assignment and chronological order



Period		1	2	3	4	5	6	7	
		young	old	young	old	young	old	young	
Cohort I	Lifecycle	1	1	:	2	3	3		
(8 subjects)	Group	A or B (random)		A or B (random)		A or B (random)			Etc.
		old	young	old	young	old	young	old	
Cohort II	Lifecycle		:	1	:	2	3	3	
(& subjects)	Group	A or B	A or B (I	random)	A or B (I	random)	A or B (I	andom)	

Bibliography

- Agnello, L. and L. Schuknecht (2011). Booms and busts in housing markets: Determinants and implications. *Journal of Housing Economics* 20(3), 171–190.
- Andrew, M. and G. Meen (2003). House price appreciation, transactions and structural change in the british housing market: A macroeconomic perspective. *Real Estate Economics* 31(1), 99–116.
- Armona, L., A. Fuster, and B. Zafar (2019, 1). Home price expectations and behaviour: Evidence from a randomized information experiment. *Review of Economic Studies* 86(4), 1371–1410.
- Bao, T., E. Halm, C. Noussair, and Y. Riyanto (2018). Ceo incentives and stock price dynamics: An experimental approach. Working Paper.
- Bao, T. and C. Hommes (2019). When speculators meet constructors: Positive versus negative feedback in experimental housing markets. *Journal of Economic Dynamics and Control 107.*
- Berkovec, J. A. and J. L. Goodman (1996). Turnover as a measure of demand for existing homes. *Real Estate Economics* 24(4), 421–440.
- Bernasconi, M. and O. Kirchkamp (2000). Why do monetary policies matter? an experimental study of saving and inflation in an overlapping generations model. *Journal of Monetary Economics* 46(2), 315–343.
- Caginalp, G., D. Porter, and V. Smith (1998). Initial cash/asset ratio and asset prices: An experimental study. Proceedings of the National Academy of Sciences of the United States of America 95(2), 756–761.
- Caginalp, G., D. Porter, and V. Smith (2001). Financial bubbles: Excess cash, momentum, and incomplete information. *Journal of Psychology and Financial Markets* 2(2), 80–99.
- Camera, G., C. Noussair, and S. Tucker (2003). Rate-of-return dominance and efficiency in an experimental economy. *Economic Theory* 22(3), 629–660.
- Claessens, S., M. A. Kose, and M. E. Terrones (2012). How do business and financial cycles interact? Journal of International Economics 87, 178–190.
- Claessens, S., M. Terrones, and M. A. Kose (2009). What happens during recessions, crunches and busts? *Economic Policy* 24 (60), 653–700.

- Crockett, S., Y. Izhakian, and J. Duffy (2018). An Experimental Test of the Lucas Asset Pricing Model. *The Review of Economic Studies* 86(2), 627–667.
- Davis, M. A. and F. Ortalo-Magne (2011). Household expenditures, wages, rents. *Review* of *Economic Dynamics* 14(2), 248–261.
- Duffy, J. (2016). Macroeconomics: A survey of laboratory research. in: J.H. Kagel and A.E. Roth (Eds.), Handbook of Experimental Economics, Princeton: Princeton University Press 2, 1–90.
- Dufwenberg, M., T. Lindqvist, and E. Moore (2005, December). Bubbles and experience: An experiment. American Economic Review 95(5), 1731–1737.
- Eckel, C. C. and S. C. Füllbrunn (2015). That she blows? gender, competition, and bubbles in experimental asset markets. *American Economic Review* 105(2), 906–920.
- Fenig, G., M. Mileva, and L. Petersen (2018). Deflating asset price bubbles with leverage constraints and monetary policy. *Journal of Economic Behavior & Organization 155*, 1–27.
- Fenig, G. and L. Petersen (2017). Distributing scarce jobs and output: experimental evidence on the dynamic effects of rationing. *Experimental Economics 20*, 707–735.
- Fischbacher, U. (2007). z-tree: Zurich toolbox for ready-made economic experiments. Experimental Economics 10(2), 171–178.
- Galí, J. (2014). Monetary policy and rational asset price bubbles. American Economic Review 104(3), 721–52.
- Greiner, B. et al. (2004). The online recruitment system orsee 2.0-a guide for the organization of experiments in economics. University of Cologne, Working paper series in economics 10(23), 63–104.
- Haruvy, E. and C. N. Noussair (2006). The effect of short selling on bubbles and crashes in experimental spot asset markets. *The Journal of Finance* 61(3), 1119–1157.
- Henderson, J. V. and Y. M. Ioannides (1983). A model of housing tenure choice. American Economic Review 73(1), 98–113.
- Hens, T. and S. C. Steude (2009). The leverage effect without leverage. Finance Research Letters 6(2), 83 – 94.

- Holt, C. A., M. Porzio, and M. Y. Song (2017). Price bubbles, gender, and expectations in experimental asset markets. *European Economic Review 100*, 72 – 94.
- Hommes, C. (2020). Behavioral and Experimental Macroeconomics and Policy Analysis: A Complex Systems Approach. *Journal of Economic Literature (forthcoming)*.
- Hort, K. (2000). Prices and turnover in the market for owner-occupied homes. Regional Science and Urban Economics 30(1), 99–119.
- Hoshihata, T., R. Ishikawa, N. Hanaki, and E. Akiyama (2017, November). Flat Bubbles in Long-Horizon Experiments: Results from two Market Conditions. GREDEG Working Papers 2017-32, Groupe de REcherche en Droit, Economie, Gestion (GREDEG CNRS), Universite Cote d'Azur, France.
- Huber, S. J. (2018). Preference for housing services and rational house price bubbles. CeNDEF Working paper 18-07, University of Amsterdam.
- Iacoviello, M. (2005). House prices, borrowing constraints, and monetary policy in the business cycle. American Economic Review 95(3), 739–764.
- Igan, D. and P. Loungani (2012). Global housing cycles. IMF Working Paper 12/217.
- Ikromov, N. and A. Yavas (2012). Asset characteristics and boom and bust periods: An experimental study. *Real Estate Economics* 40(3), 603–636.
- IMF (2003). When bubbles burst. World Economic Outlook, (Chapter 2, April).
- Jaworski, T. and E. O. Kimbrough (2016). Bubbles, crashes, and endogenous uncertainty in linked asset and product markets. *International Economic Review* 57(1), 155–176.
- King, R. G., C. I. Plosser, J. H. Stock, and M. W. Watson (1991, September). Stochastic Trends and Economic Fluctuations. *American Economic Review* 81(4), 819–40.
- Kirchler, M., C. Bonn, J. Huber, and M. Razen (2015). The "inflow-effect"—trader inflow and price efficiency. *European Economic Review* 77, 1 19.
- Kirchler, M., J. Huber, and T. Stöckl (2012). That she bursts: Reducing confusion reduces bubbles. American Economic Review 102(2), 865–883.
- Kopanyi-Peuker, A. and M. Weber (2021). Experience Does Not Eliminate Bubbles: Experimental Evidence. *The Review of Financial Studies* 34, 4450–4485.

- Kose, T. (2015, 01). Price convergence and fundamentals in asset markets with bankruptcy risk: an experiment. International Journal of Behavioural Accounting and Finance 5, 242.
- Lei, V. and F. Vesely (2009). Market efficiency: Evidence from a no-bubble asset market experiment. *Pacific Economic Review* 14(2), 246–258.
- Leung, C. (2004). Macroeconomics and housing: a review of the literature. Journal of Housing Economics 13(4), 249–267.
- Lim, S. S., E. C. Prescott, and S. Sunder (1994). Stationary Solution to the Overlapping Generations Model of Fiat Money: Experimental Evidence, pp. 75–97. Heidelberg: Physica-Verlag HD.
- Marimon, R. and S. Sunder (1993). Indeterminacy of equilibria in a hyperinflationary world: experimental evidence. *Econometrica* 61(5), 1073–1107.
- Noussair, C., S. Robin, and B. Ruffieux (2001). Price bubbles in laboratory asset markets with constant fundamental values. *Experimental Economics* 4(1), 87–105.
- Noussair, C. N. and S. Tucker (2016). Cash inflows and bubbles in asset markets with constant fundamental values. *Economic Inquiry* 54(3), 1596–1606.
- Ozimek, A. (2014). Sticky rents and the cpi for owner-occupied housing. SSRN Working Paper.
- Piazzesi, M., M. Schneider, and S. Tuzel (2007). Housing, consumption and asset pricing. Journal of Financial Economics 83(3), 531–569.
- Plott, C. R. and V. L. Smith (2008). Handbook of experimental economics results. 1.
- Porter, D. P. and V. L. Smith (1995, October). Futures contracting and dividend uncertainty in experimental asset markets. *Journal of Business* 68(4), 509–541.
- Razen, M., M. Kirchler, and J. Huber (2017). Cash inflow and trading horizon in asset markets. *European Economic Review 92*, 359 – 384.
- Schularick, M. and A. M. Taylor (2012, April). Credit booms gone bust: Monetary policy, leverage cycles, and financial crises, 1870-2008. American Economic Review 102(2), 1029–61.

- Smith, V. L., G. L. Suchanek, and A. W. Williams (1988). Bubbles, crashes, and endogenous expectations in experimental spot asset markets. *Econometrica* 56(5), 1119–1151.
- Smith, V. L., M. van Boening, and C. P. Wellford (2000). Dividend timing and behavior in laboratory asset markets. *Economic Theory* 16(3), 567–583.
- Stein, J. C. (1995). Prices and trading volume in the housing market: A model with down-payment effects. The Quarterly Journal of Economics 110(2), 379–406.
- Stöckl, T., J. Huber, and M. Kirchler (2010). Bubble measures in experimental asset markets. *Experimental Economics* 13(3), 284–298.
- Weber, M., J. Duffy, and A. Schram (2018). An experimental study of bond market pricing. The Journal of Finance 73(4), 1857–1892.
- Weitzel, U., C. Huber, J. Huber, M. Kirchler, F. Lindner, and J. Rose (2019, 08). Bubbles and Financial Professionals. *The Review of Financial Studies*. hhz093.